Asian Review of Accounting



Political connections, board ethnicity and value relevance in Mauritius

Journal:	Asian Review of Accounting
Manuscript ID	ARA-10-2022-0238.R2
Manuscript Type:	Research Paper
Keywords:	Political connections, board of directors, Ethnicity, Value relevance



Political connections, board ethnicity and value relevance in Mauritius

Abstract

Purpose - The paper aims to investigate the impact of political connections and board ethnicity on the value relevance of earnings and book value in Mauritius.

Design/methodology/approach - This study is based on a sample of 541 Mauritian listed firm-year observations for 2001-2016. Financial and board diversity data have been collected using the listed firms' annual reports and from reports published by the Stock Exchange of Mauritius. Political connection data was derived from the directory of Chief of State and Cabinet members. The research hypotheses were empirically tested using a modified Ohlson (1995) price model.

Findings - This study shows that political connections negatively impact the value relevance of earnings and book value. We find that firms with Franco-Mauritian directors will constrain political connections' negative impact. We find contrasting results for Indo-Mauritian directors since they form an integral part of the government in Mauritius.

Originality - This study contributes to the scarce accounting literature in Mauritius. Firstly, no study has investigated the relationship between the value relevance of accounting information and political connections in Mauritius. Secondly, Mauritius' capital market is dominated by a non-indigenous ethnic group, Franco-Mauritians, who remains the economic elite. Hence, Mauritius presents an opportunity to bring forth another important aspect in the capital market and corporate governance; diversity on the board of directors. Therefore, the study extends to the political connections and board diversity literature.

Keywords - political connections, board diversity, ethnicity, value relevance, Mauritius.

JEL classifications: G3, G14, M14, M41

Paper type Research paper

1. Introduction

Several studies have examined the economic consequences of political connections in capital markets. Studies on political connections center, but are not limited to, the quality of accounting information (Chaney *et al.*, 2011), audit fees (Gul, 2006), capital and resources allocations (Johnson and Mitton, 2001), financial analysts (Chen *et al.*, 2010; Gist and Abdul Wahab, 2021), firm performance (Fisman, 2001), auditor choice (Guedhami *et al.*, 2014; Tantawy and Moussa, 2023), cost of equity capital (Boubakri *et al.*, 2012) and corporate lending (Khawaja and Mian, 2005).

The findings of these studies are rather mixed, suggesting that political connections could be detrimental or not to capital markets. On one hand, these studies suggest that political connections or politically connected firms are rent-seekers and promote cronyism and nepotism. On the other hand, conventional wisdom tells us that being connected would provide capital funding via grants and increase performance (Fisman, 2001). In addition, the stream or resources from the income for connected firms are relatively uncertain, volatile, and require higher scrutiny.

Mauritius's political connections and earnings quality studies are limited and in the infancy stage. The extant literature in Mauritius has primarily focused on the implementation of the code of corporate governance (Mahadeo and Soobaroyen, 2013), the investigation of board diversity (gender, age, and education), and corporate governance (Munisi and Randoy, 2013), firm performance (Mahadeo *et al.*, 2012), and a case study investigating the role of shareholders activism in Mauritius (Beebeejaun and Koobloll, 2018).

We extend these studies on several grounds. First, none of the above studies has examined political connections, an essential feature in a developing capital market like Mauritius. Given the interest, we investigate whether political connections affect the

Page 3 of 114

value relevance of accounting information in Mauritius, as no study, to date, has investigated this relationship. Mauritius is an emerging economy in the African region, and the findings provide insight into the nature of connections in the country. Given the unique background that shapes its capital market, Mauritius provides an interesting avenue. We are driven by several high-profile ongoing cases surrounding politicians and heads of government. For example, soon after the December 2014 elections, the previous Head of Government, Dr. Navinchandra Ramgoolam, was arrested based on money laundering charges when two safes with billions of rupees and dollars were discovered at his residence (Wan, 2015).¹ In March 2016, the Prime Minister, Sir Anerood Jugnauth, asked the Minister of Environment to step down after bribery allegations were made against him (Reuters, 2016). These cases indicate that political connections are prevalent and influential in Mauritius, questioning the linkage between political board members and the value of published accounting numbers.

Secondly, the Mauritius capital market is dominated by a non-indigenous ethnic group, Franco-Mauritians, who remain the economic elite (Carroll and Carroll, 2000). Hence, Mauritius presents an opportunity to bring forth another important aspect of capital markets and corporate governance; diversity on the board of directors. Board diversity is important in promoting good governance (Carter *et al.*, 2010). The guiding ethics behind board diversity is to promote governance and better monitoring by providing helpful insight, obtaining external resources, expertise, and exercising diligence (Simkins and Carter, 2003; Srinidhi *et al.*, 2011; Tee, 2019). Consequently, in theory, these traits enhance firm value and the value relevance of accounting information (Ntim, 2013). However, this is not always true when coupled with a nation's socio-economy disparity (Darga and Joomun, 2005; Mishra and Jhunjhunwala, 2013).

A distinctive feature of the Mauritius capital market is that it is dominated by white Franco-Mauritians (Mauritians of White-French descendants), who only constitute around two percent of the total population. The other ethnic group, Indo-Mauritians, does not dominate the capital market despite forming almost 68 percent of the population (Srebrnik, 2002; Bunwaree and Kasenally, 2005). This 'disparity' or 'mismatch' between the ethnic groups and Franco-Mauritians' dominance in the capital market presents interesting avenues for research in this area. This landscape is not unique to Mauritius, as other countries, such as Malaysia, experience a similar situation. However, the dominance of the non-majority Chinese-Malaysian is not as prevalent as Franco-Mauritians in Mauritius. Mauritius has not had a question regarding ethnicity in the national census since 1972; however, Indo-Mauritians consist of two-thirds of the total population (CIA World Factbook, 2021), and several researchers refer to the following statistics: the Indo-Mauritians 68% (Hindus 52% and Muslims 16%); Creoles Mauritians 28%; Sino-Mauritians 3% and Franco-Mauritian less than 2% of the population (Srebrnik, 2002; Bunwaree and Kasenally, 2005; Salverda, 2015). We opted for these two board ethnicity measures as they represent 'extremes' (Salverda, 2015) as one group (Franco-Mauritians) is dominant in the capital market (in this case, board participation). At the same time, Indo-Mauritians are a minority in the capital market, although they have the largest population. These differences present us with competing hypotheses to test.

We predict that white Franco-Mauritian will mitigate the negative impact of political connections on the value relevance of book value and earnings. The premise of our argument is simple. The white Franco-Mauritians has been a dominant player in the capital market, providing diverse expertise, know-how, and, more importantly, experience. Their presence on the board of directors provides these characteristics that could offer a monitoring role and increase earnings quality. The white Franco-Mauritian

are not known to be active participants in the political scene in Mauritius. One would say that their dominance in the capital market presents a form of check and balance to Mauritius's politicians or political connections. Since white Franco-Mauritians are not active in politics, their survival in the capital market could signal their ability to tap into the external capital market and demonstrate the ability to sustain themselves without government assistance (Salverda, 2015).²

In contrast, we predict that political connections will negatively impact the relevance of book value and earnings for Indo-Mauritians. The existing literature supports this view. Although dominant in the political scene, Indo-Mauritians are less prominent in the capital market. They often use their presence in the capital market as a symbol, mainly for attracting funding or government grants (Salverda, 2015). For example, the presence of Malay directors in Malaysia during the early stages of independence was largely ceremonial and served as an indicator of exposing government servants to the capital market (Abdul Wahab *et al.*, 2018).

Based on a sample of 541 firm-year observations of Mauritian listed firms (SEM) for 2001-2016, we find that both the book value and earnings are value relevant in the capital market. As predicted, we find the value relevance of book value is lower for politically connected firms. This finding is consistent with the argument that politically connected firms have lower earnings quality and create uncertainty in the market. Furthermore, the negative impact of political connections is compounded by the recent high-profile cases involving local politicians, as mentioned above.

The results confirm our predictions: the presence of Franco-Mauritian directors mitigates the negative impact of political connections on the value relevance of earnings. This suggests that Franco-Mauritian directors mitigate agency issues and monitor and perform checks and balances on political connections. In support of our arguments, we

find that Indo-Mauritian directors increase the negative impact of political connections on value relevance. The results are unsurprising since Indo-Mauritians dominate the government as politicians and officers. After controlling the selection test and several additional analyses, the results remain qualitatively similar.

The findings of this study contribute to the literature on political connections, ethnicity, and value relevance in the following ways. First, the paper extends the evergrowing literature on political connections, especially in Africa. Given Mauritius's unique political, legal, and social backgrounds relative to other African countries, the findings of this study act as a catalyst for further political connection studies.

The results of this study present another view on how ethnic groups interplay with the premise of political connections. The institutional background of Mauritius allows us to examine the different dimensions of political connections in the capital market. Second, the study tackles ethnicity and provides an exciting outcome on how ethnic groups play their role in developing the capital market. Third, the findings of this study contribute to the understanding of how the various ethnic groups in a capital market play a role.

The paper is organized as follows. Section 2 summarizes the political and social background in Mauritius. Section 3 provides the rationale behind the hypotheses developed. Section 4 presents the methodology and sample selection. The results and robustness tests are discussed in Section 5, while section 6 discusses the additional analysis. Section 7 concludes the paper.

2. Institutional background

2.1. Political Economy in Mauritius

Since 1810, British governors and officials administered the island of Mauritius. In the 1960s, the negotiations for independence started, and elections were held in 1967.

Consequently, Mauritius became a fully-fledged democracy and a Commonwealth member on 12 March 1968. The independent state was then run by the Labour Party and two smaller parties: The Independent Forward Bloc and the Muslim Action Committee – two ethnic-based parties. Since 1979, the elections have resulted in coalition governments with the same political parties rotating from the government to the opposition. Therefore, community-based favoritism and nepotism remain a source of public frustration even though corruption is not customary per regional standards.³

The political structure in Mauritius was adapted from the British Empire and implemented after independence in 1968.⁴ In line with its globalization strategy, since the 1980s, the government of Mauritius has taken the role of a business facilitator to create a positive business environment on the political, economic, and social levels. The aim is to nourish and expand the private sector. Furthermore, the government has controlling shares in numerous listed firms: Air Mauritius – the national airline; State Bank of Mauritius; and Mauritius Telecom – the telecommunication company. The government also invests in various corporations via its investment entity – the State Investment Corporation. The board of directors' positions are primarily allocated to senior government officials, with the chairperson nominated by the ruling government. The chairpersons are rotated once there is a change in government. The new ruling party has the authority to remove a chairperson nominated by the previous government and replace him/her with a new chairperson fitting the ideologies of the new ruling party.

However, Franco-Mauritians have retained major land ownership; over time, they have created family holdings to manage those lands (Salverda, 2013). The family holdings still exist and are majorly quoted on the SEM. The boards of directors of those listed firms are composed of directors drawn from the same small and tightly-knit white

Franco-Mauritian business community. The directors are dominant male directors appointed to numerous boards for decades (Mahadeo *et al.*, 2012).

Mauritius has no indigenous population but is known for its cultural diversity. As Mahadeo and Soobaroyen (2013) mentioned, Mauritius has a dual colonial past being both an ex-French and British colony, and therefore its population came mainly from European colonialism and immigration. Also, Brautigam (2009) argues that even though the population originates from the world's four corners – Europe, Africa, India, and China- it has retained close links with French and British traditions.

In 1835, Mauritius' geographical setting was dominated by sugar cane plantations showing early signs of a one-crop economy that relied on slave labor (Holmberg, 1962). However, after the abolition of slavery in the British Empire in 1834, the demand for sugar cane plantations' labor increased, especially with preferential prices being introduced by Britain on Mauritian sugar. The need for such labor led to the "Great Experiment" initiated by the British government. More than 450,000 laborers were expatriated from India to Mauritius, with the status of indentured laborers working in Mauritius' sugar cane fields. Most of today's Indo-Mauritians, of Hindu or Muslim faiths, are descendants of these indentured laborers, while a minority came freely as traders or educators (Chiriyankandath, 2009). Two-thirds of these Indo-Mauritians remained permanently on the island, while one-third returned to their home country or migrated to other British colonies (Aapravasi Ghat Trust Fund, 2015).

2.2. Social background and its impact on financial reporting

Indeed, while the Mauritian population is diverse, the largest firms' property and business ownership remain concentrated and traced back to the heritage left by colonization (Lange, 2003). The World Bank (2010) reports that white Franco-Mauritian families control between 5 and 7 large listed firms. These families, which benefited mainly from

the sugarcane industry, consequently invested massively in other local industries. This can also be observed in the directors' composition: most of the directors are white Franco-Mauritian directors, while some minority directors come from the Creole community, the Indo-Mauritian community, and the Sino-Mauritian community. This highlights the discrepancy between the composition of the Mauritian population and the diversity of the boards of directors. While many directors appear to have been appointed for their knowledge, skills, and expertise, some directors are appointed due to connections to family and majority shareholders. Therefore, those family-dominated management structures have a definite impact on the boards of SEM-listed firms.

Furthermore, the majority shareholder usually appoints non-executive directors, which questions the independence of those nominated. This again creates an interesting setting to probe the influence of board diversity on published accounting information's value relevance.

3. Hypotheses development

3.1. Value relevance in the Stock Exchange of Mauritius (SEM)

The accounting literature defines value relevance as the association between company values and accounting numbers (Ohlson, 1995; Francis and Schipper, 1999; Barth *et al.*, 2001; Kothari, 2001; Beisland *et al.*, 2010). Value relevance research investigates the usefulness of reported numbers to accounting information users and current and potential investors.

Ball and Brown (1968), Beaver (1968), and Beaver *et al.* (1980) explore a plausible association between accounting variables and share prices. Beaver (2002) explains that value relevance research examines how accounting variables relate to a

dependent variable based on security price. Negakis (2005) believes that this research aims to assess how helpful accounting information is for users, like investors.

Indeed, investors widely use financial reports to assess the market value of firms for investment decisions (Adedeji and Kajola, 1999). Consequently, it can be seen that the main objective of financial reporting is to provide equity valuation information to potential investors. This objective is empirically analyzed by value relevance research. This justifies the study of the value relevance of accounting information in Mauritius; subsequently leading to the first hypothesis:

H₁: Accounting information produced by SEM-listed firms is value relevant.

Badu and Appiah (2018) studied the value relevance of accounting information of public firms using Ohlson's price model (1995) applied on the Ghana Stock Exchange for 2005-2014 using a sample of 224 firm-year observations. They find a significant and positive relationship between earnings and book values of equity to stock prices. Furthermore, the same conclusion is reached by Diftar and Elkalla (2019), who examined the value relevance across the different countries of North Africa and the Middle East from 2007 to 2016.

3.2. Political connections and value relevance

According to Faccio (2006), a firm is politically connected if one of the firm's largest shareholders or senior executives is a member of parliament (MP), a minister, or the head of government, or closely related to a top official. A large shareholder is defined as anyone who controls a minimum of 10% of the shareholders' votes directly or indirectly (Faccio 2006). Faccio (2006) also includes the political connections through a relative, whereby one relative of a head of state or minister is a large shareholder or a management executive. Firms are also classified as politically connected if: (i) A head of state or one

Page 11 of 114

Asian Review of Accounting

of his/her relatives was also a top executive or MP during the study period; (ii) A current government minister or MP was a senior executive or significant shareholder of a company in the past; (iii) A top executive or substantial shareholder is a politician of another country; (iv) A substantial shareholder or senior executive is known to be associated with a political party whether the ruling parties or opposition parties (Civilize *et al.*, 2015). Other researchers classified state ownership as political ties with firms and special shares held by the government (Jones *et al.*, 1999; Bushman *et al.*, 2004; Nee *et al.*, 2007; Hanousek *et al.*, 2007).

There are many ways for firms to obtain benefits from political connections. Such connections can lower tax burdens (Adhikari *et al.*, 2006), increase the preferential treatment to financing (Dinc, 2005) and political bailouts in the event of financial distress (Faccio *et al.*, 2006), and result in a higher allocation of government investment during a financial crisis (Johnson and Mitton 2003). Chaney *et al.* (2011) suggest that political connections should increase earnings quality due to heightening media scrutiny, leading to increased firm monitoring. With increased scrutiny, better access to resources, and enhanced monitoring due to the public or state interest, these connected firms should have higher earnings quality relative to non-connected firms.

Batta *et al.* (2014) investigate whether a country's degree of expropriation risk could amount to a positive rapport between political connections and earnings quality. Batta *et al.* (2014) argue that countries with high expropriation risks will lower their quality of earnings, but not for politically connected firms since the risk is lower due to connections. Even though Mauritius has a low ranking in terms of expropriation risk, in 2014, the government took over the BAI Group, a leading conglomerate in Mauritius. The group's assets were seized and sold to local firms, while some were retained under the government's control. Moreover, Fung *et al.* (2015) similarly claim that the duration

of a connection could suggest that the connected firm may self-sustain into the future and thus could increase earnings. Political connections create some form of certainty, especially in dealing with resources. The above suggests a positive affiliation between political connection and earnings quality; hence a positive affiliation between political connection and the value relevance of accounting information is a component of earnings quality. Conventional wisdom suggests that political connections or relationship-based economies provide an advantage to the connected firms to capital (Fisman, 2001) and information from the government on policy changes. Such advantages would improve the market value of connected firms relative to unconnected firms.

A classic example was from the work of Johnson and Mitton (2003) when they examined the impact of capital control on connected firms in Malaysia during the Asian Financial Crisis of 1998-1999. They uncovered that capital control restricted the outflow of funds and assisted or propped the connected firms during the crisis. Although the capital control period indicates that the connected firms were operating inefficiently before the crisis, such policy (e.g., restriction of the flow of funds) increases the value of connected firms.

Alternatively, existing literature suggests political connections could harm the firm's value and earnings quality. Political connections create agency costs between agent and principal and minority and majority shareholders (Faccio, 2006). The agency costs are owed to the high rent-seeking activities by the directors with political connections and the low level of transparency (Ball *et al.*, 2003). The high agency cost increases information asymmetry and creates uncertainty, decreasing earnings quality. Chaney *et al.* (2011) offer three reasons to contribute to a negative relationship between political connections and the quality of earnings. First, insiders of connected firms could hide, obscure, or delay reporting the advantages received to deceive investors. Second, Chaney

Page 13 of 114

et al. (2011) argue that the connected firms do not worry about the quality of accounting information as politicians shield them; and the third argument is that firms with poor earnings quality are more inclined to establish political connections. Therefore, the political connection appears to have an ambiguous impact on investors, and thus, this leads to the following research hypothesis:

H₂: Political connections weakens the value relevance of book value and earnings

The evidence examining the relationship between political connections and earnings quality is mixed. Chen *et al.* (2010) investigate the relationship between analyst forecast accuracy and political connections and find a negative relationship. They argue that uncertainty in the revenue-generation process of connected firms is highly questionable. Gul (2006) investigates the relationship between audit fees and political influences and finds a positive relationship. Gul argues that auditors perceived connected firms as risky due to the high agency costs. Faccio (2006) shows no significant price effect when a politician is appointed to corporate boards, yet the share price increases when an executive enters politics. On the same line, many authors establish that earnings quality is systematically lower for firms with political connections (Chaney et al., 2011; Riahi-Belkaoui, 2004), and Leuz and Oberholzer-Gee (2006) find a reduction of foreign financing in politically-connected firms. Fisman (2001) shows that investors reacted to every rumor about the health of President Suharto. Thus, the share returns of politically connected Indonesian firms are noticeably lower than those of unconnected firms. The same is confirmed by Bertrand et al. (2004); Fan et al. (2007); Li et al. (2008); and Civilize et al. (2015) in the French, Chinese, and Thai contexts, respectively.

3.3. Political connections, board ethnicity, and value relevance

The agency theory suggests that board diversity increases the independence of the board

as the differences in gender, ethnicity, and background trigger additional questioning of executive's decisions (Arfken *et al.*, 2004; Johnston and Malina, 2008; Carter *et al.*, 2010; Lincoln and Adedoyin, 2012; Abdullah, 2013; Triana *et al.*, 2014). As the human capital theory mentions, board diversity brings increased creativity and innovation, new ideas and perspectives, and new skills and knowledge, which will help decision-making (DiStefano and Maznevski, 2000; Baranchuk and Dybvig, 2009; Luckerath-Rovers, 2013). Board diversity can bring the company closer to its stakeholders and create a more general responsibility towards society (Van Der Walt and Ingley, 2003).

Consequently, several interdisciplinary theories strongly indicate a link between a firm financial performance and board diversity. Human capital and resource dependence theories suggest a positive relationship between firm performance and board diversity (Carter *et al.*, 2010). Yet, the agency theory advises that the nature of the relationship might not be explicit even though proper management monitoring may drive higher firm value (Ntim, 2013).

The appointments of directors based on ethnic grounds could be for several reasons. Resource dependency theory suggests that such appointments are based on human and social capital needs. These appointments are usually based on special skills only obtained by certain ethnic groups (human capital) or the needs of firms to seek economies of scale via networking (social capital). Many studies state that the board of directors remains one of the utmost crucial subsets within an organization (Lipton and Lorsch, 1992; Sonnenfeld, 2002; Bart and McQueen, 2013) as it performs many strategic functions within the organization (Bilimoria and Piderit, 1994; Lincoln and Adedoyin, 2012; Dale-Olsen *et al.*, 2013; Ntim, 2013). In addition, Agarwal and Knoeber (2001) argue that firms acquire different advantages and resources when they appoint ethnic

Asian Review of Accounting

minority directors, and boards with several dealings with the government or face government intervention and regulation are more likely to appoint the same directors.

The studies demonstrate that ethnic diversity enhances the corporate governance element of board independence, monitoring executives' decision-making, and acquiring resources, consequently improving the firm's value (Ntim, 2013). Furthermore, ethnic directors provide unique information to the board, which provides better information to the executives and improves decision-making (Abdullah, 2013).

Oppositely, Goodstein *et al.* (1994) explored board diversity's impact on decisions involving strategic changes, and they concluded that homogenous boards tend to initiate strategic changes more often than diverse boards. This implies that diverse boards might negatively impact firm valuation. Several researchers explain that directors from ethnic minorities might be chosen by tokenism; to tick a box imposed by the government (Hillman *et al.*, 2007; Abdullah, 2013; Ntim, 2013). Finally, a diverse board may increase conflicts between board members as they bring their interests and loyalties to the board (Roberson and Park, 2007; Wellage and Locke, 2013), preventing board cohesion (Goodstein *et al.*, 1994; Lincoln and Adedoyin, 2012).

Preceding studies have indicated that political affairs may impact board composition (Agarwal and Knoeber, 2001; Fan *et al.*, 2007; Chen *et al.*, 2010), as directors with political connections bring several advantages. Agarwal and Knoeber (2001) suggest that outside directors who have political connections can assist the firm in navigating political affairs by using their abilities to anticipate or influence government actions. These abilities can come from their previous involvement in government or political parties and their experience with legal proceedings against the government. Domadenik *et al.* (2016) mention that a government might use its power illegitimately to appoint people without adequate competencies to boards of state-owned firms as long as

they belong to the political network. The appointed directors are positioned as a special purpose vehicle to extract rent or are being rewarded for previous political engagements.

From an accounting point of view, diverse boards, especially ethnically diverse boards, are expected to lead to better performance and higher earnings quality through enhanced corporate governance and accountability (Srinidhi *et al.*, 2011; Tee, 2019). Furthermore, Kim *et al.* (2013) studied the effect of boards with diverse political ideologies on firm performance. They concluded that boards with political diversity are linked with higher firm performance (Liang *et al.*, 2021). While the literature in this field is scarce, it might apply in Mauritius. Indeed, political connection measures the relationship between firms and political parties. However, in Mauritius, adherence to political parties or simply maintaining a relationship with a specific party is closely related to the ethnic group to which the person belongs.

We argue that board ethnicity, our proxy for board diversity, would moderate the negative impact of political connections on the value relevance of earnings and book value. The premise of the moderating variable is simple, and one would expect the positive impact of board diversity would mitigate the negative relationship between political connections and the value relevance of earnings and book value.

However, the institutional background in Mauritius would provide a different story. Our study focuses on two ethnic groups: Franco-Mauritians, primarily white and descended from French and British colonizers, and Indo-Mauritians. They originated from India and reached Mauritius as indentured laborers, merchants, traders, and highprofile teachers and officers under British rule. The Franco-Mauritians group, representing a minority ethnic group in Mauritius, dominates the local capital market. However, despite having massive control in the capital market, they are hardly present as officers or local representatives in the government. We expect that the negative impact of

political connections on the value relevance of earnings and book value is weakened by the presence of Franco-Mauritian directors. This scenario presents a 'check and balances' to the government's influence in the capital market. In addition, their presence in the capital market signals their ability to tap into the external market. Based on the above, we predict the following hypothesis:

*H*₃: *The negative impact of political connections on value relevance will be weakened by Franco-Mauritian directors*

The appointment of Indo-Mauritian directors who dominate the political scene could be seen as ceremonial rather than as playing an active role in monitoring. In contrast, we expect that the presence of Indo-Mauritian directors will not mitigate the negative impact of political connections on the value relevance of earnings and book value. Carroll and Carroll (2000) highlight that the government deliberately expanded public employment in the 1970s by hiring unskilled and unemployed, mostly Indo-Mauritians. This has escalated to ministries and parastatal organizations (Carroll and Carroll, 2000). In anecdotal evidence supplied by Carroll and Carroll (2000), employment in these institutions is presented as a reward to supporters and clients. Hence, in line with prior theories and studies, the following is hypothesized:

 H_4 : The negative impact of political connections on value relevance will be heightened by Indo-Mauritian directors

Several studies have examined this relationship in a similar setting. Studies have documented evidence in a similar setting. Abdul Wahab *et al.* (2015) investigate the relationship between culture and analyst forecast accuracy in Malaysia. According to them, Bumiputera directors, the largest ethnic group in Malaysia but not as prominent in

the capital market, have lower governance levels, less disclosure capacity, and more inherent risk and inefficiency due to preferential treatment. This will lead to an increase in agency costs. Based on this premise, they argue that the increased agency cost will increase information asymmetry, impacting analysts' forecasts. Abdul Wahab *et al.* (2015) find results that support their arguments.

4. Methodology

4.1. Sample and Data

The initial population of the study is the firms listed on the SEM from 2001 to 2016. The number of firms listed during the sample period is 43, resulting in 688 firm-year observations. We exclude financial firms as they are subject to different accounting regulations and risk structures. We exclude the three (3) financial firms, which result in 48 firm-year observations. Due to missing financial data, we exclude 99 firm-year observations. Hence, the final sample for this study is 541 firm-year observations, as shown in Table 1.

(Table 1 here)

Financial data has been collected using the Thomson Reuters DataStream database, Bureau van Dijk's Osiris database, and mostly the listed firms' annual reports. The stock prices were gathered from reports published by the Stock Exchange of Mauritius. Political connection data has been derived based on the directory of Chief of State and Cabinet members of Foreign Countries, compiled by the Central Intelligence Agency of the United States of America. Finally, data used to identify board diversity was gathered primarily using the firms' annual reports and company websites.

The value relevance of accounting information is indicated by a statistical association between stock prices or stock returns and a set of accounting information

(Beaver, 2002; Liu and Liu, 2007). Indeed, value relevance research intends to provide empirical data displaying the degree to which accounting numbers shape equity valuation and subsequently aims to assess the usefulness of these numbers to investors' decisionmaking.

Therefore, value relevance is usually tested using regression analysis: price regression or return (price change) regression. Early research in value relevance started with price regression, examining the relationship between the book value and the market value of equity. It is usually denoted as follows:

$$P^{it} = \beta_o + \beta_I B V^{it} + \varepsilon^{it}$$

Where:

 P^{it} = Market value or price per share of firm *i* at time *t* BV^{it} = Book value of total equity or Book Value per share of equity of firm *i* at time *t*

Furthermore, Ohlson (1995) shows that firm's value is a linear function of its book value, earnings and other relevant information (Collins *et al.*, 1997; Kim and Kross, 2005). Hence, the formal regression of this study is as per the following equation:

$$P^{it} = \beta_o + \beta_I EPS^{it} + \beta_2 BV^{it} + \varepsilon^{it}$$

e of firm *i* for the period *t*

 EPS^{it} = Earnings per share of firm *i* for the period *t*

Glasscock et al. (2021) shed light on the ongoing debate for proper scaling in empirical accounting research. They believe alternative specifications, other than scaling, may be more effective in addressing heteroscedasticity issues. Lubberink and Willett (2021) and Boonlert-U-Thai and Schaberl (2022) argue that log-linear regression models are appropriately suited and provide valid estimates of the relationship between price and

accounting data. Therefore, this study uses a log-linear model of Ohlson's (1995) model to estimate the value relevance of accounting information:

$$LnP^{it} = \beta_o + \beta_1 LnEPS^{it} + \beta_2 LnBV^{it} + \varepsilon^{it}$$

Where:

 $LnP^{it} = Log of market value or price per share of firm i at time t$

LnEPS it = Log of Earnings per share of firm i for the period t

 $LnBV^{it} = Log of Book value of total equity or Book Value per share of equity of firm$ *i*at time*t*.

In line with the log-linear regression models used in Boonlert-U-Thai and Schaberl (2022), we use the following final regressions:

 $LnP_{it} = \beta_0 + \beta_1 LnEPS_{it} + \beta_2 LnBV_{it} + \beta_3 POLCON_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 LNSIZE_{it} + \beta_7 LEV_{it} + \beta_8 MANOWN_{it} + \beta_9 INSTOWN_{it} + \beta_{10} LNAGE_{it} + \beta_{11}CG2004_{it} + \beta_{12}INDUSTRY_{it} + \beta_{13}POL^*EPS_{it} + \beta_{14}POL^*BV_{it} + \varepsilon_{it}$ (Equation 1)

 $LnP_{it} = \beta_0 + \beta_1 LnEPS_{it} + \beta_2 LnBV_{it} + \beta_3 POLCON_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 BODFRANCO*LnEPS_{it} + \beta_7 BODFRANCO*LnBV_{it} + \beta_8 BODINDO*LnEPS_{it} + \beta_9 BODINDO*LnBV_{it} + \beta_{10} POLCON*LnEPS_{it} + \beta_{11} POLCON*LnBV_{it} + \beta_{12} POLCON*LnEPS_{it} + \beta_{13} POLCON*LnBV_{it} + \beta_{14} POLCON*BODFRANCO_{it} + \beta_{15} POLCON*BODINDO_{it} + \beta_{16} POLCON*BODFRANCO*LnEPS_{it} + \beta_{17} POLCON*BODFRANCO*LnBV_{it} + \beta_{18} POLCON*BODINDO*LnEPS_{it} + \beta_{19} POLCON*BODFRANCO*LnBV_{it} + control variables + \varepsilon_{it}$

(Equation 2)

4.2. Independent variables

The political connection, $POLCON_{it}$, will be measured using Faccio's (2006) definition of politically connected firms and its methodology. In this study, the political connection variable takes the value of 1 if the firm is considered a politically connected firm; otherwise, a value of zero is granted. The methodology for identifying a politically connected firm is presented in Appendix B.

Board ethnic diversity has been measured based mainly on the ethnicity of the Mauritian population. The directors have been classified as Franco-Mauritian directors (*BODFRANCO_{it}*) and Indo-Mauritian directors (*BODINDO_{it}*). Franco-Mauritian directors are Mauritian whites, mostly descendants of French and British colonizers. In contrast, Indo-Mauritian directors originated from India and reached Mauritius as indentured laborers, merchants, and traders, and some as high-profile teachers and officers under British rule.

4.3. Control Variables

Following recommendations from the existing literature, control variables are exploited to manage the effect of firm-related variables: Size of the firm ($LNSIZE_{it}$), leverage of the firm (LEV_{it}), direct managerial ownership of the firm ($MANOWN_{it}$), institutional ownership of firm ($INSTOWN_{it}$), and lastly firm's age since incorporation ($LNAGE_{it}$). The firm's size is operationalized by the natural log transformation of total assets of firm *i* at time *t* in millions of Mauritian Rupees, while leverage is operationalized by the natural logarithm of total debt divided by total assets. Direct managerial and institutional ownership are quantified by the direct managerial ownership percentage of firm *i* and the ownership percentage by the top 5 direct institutional shareholders of firm *i*. Finally, the age of firms are calculated by the natural logarithm of the number of years since incorporation.

Industry effects are also controlled by the industry dummies, categorized into eight industries based on Mauritius's Official Market, the Stock Exchange.⁵ Following the context of Mauritius, a dummy variable representing the year 2004 (*CG2004*) has been included to capture the effect of the enactment and implementation of the Code of Corporate Governance. Please refer to Appendix A for the variables' definitions.

(Appendix A here)

5. Results

5.1. Descriptive Statistics

Table 2 shows the descriptive statistics relating to this research. Panel A shows the descriptive statistics for the different variables included in Ohlson's value relevance model: the share price (*PRICE_{it}*), the Earnings per share (*EPS_{it}*), and the Book Value of shares in Mauritian Rupees of SEM-listed firms for the period 2001 – 2016. Panel B of Table 2 reports the descriptive statistics of the independent variables: Political connection and ethnicity. Political connection (*POLCON_{it}*) takes the value of 1 if the firm is politically connected; therefore, 36.6% of the sample have some political connections per the descriptive statistics. Appendix B explains our methodology for identifying the politically connected firms in Mauritius.

(Appendix B here)

Board diversity shows that, on average, 68% of directors are white, while 32% are nonwhite. Ethnicity is further analyzed by the percentage of Franco-Mauritian directors (BODFRANCO_{it}) and Indo-Mauritian directors (BODINDO_{it}). The average percentages of BODFRANCO_{it} and BODINDO_{it} are 60.71% and 7.4%, respectively. Boards of directors are mostly composed of Franco-Mauritian directors. This is explained by Mauritius being an ex-British and ex-French colony.

(Table 2 here)

5.2. **Correlations**

Table 3 presents the Pearson and Spearman correlations for the variables included in Ohlson's model of value relevance, the experimental variables applicable to the context of Mauritius, and the control variables. They both yield similar results; however, the results from the Spearman correlations are more significant. The significant correlations support the hypotheses that political connections and the ethnic diversity of the board of directors influence the dependent variable. Both Pearson and Spearman correlations indicate that it will be relevant to test the hypotheses on the value relevance of accounting e) information.

(Table 3 here)

5.3. Univariate

5.3.1. Political connections

Table 4 exhibits the results from the univariate analysis for the test variables relating to politically connected and politically unconnected firms. Panel A of Table 4 shows that the share price varies for both groups; politically connected firms demonstrate higher share prices (*PRICE_{it}*) than non-politically connected firms (Rs 67.121 as opposed to Rs 58.589). The same is observed for the log variable $LnPRICE_{it}$. The earnings per share

 (EPS_{it}) and the book value of equity (BV_{it}) and their associated log variables are higher for politically connected firms than politically unconnected firms, which suggests that the value relevance of accounting information is potentially different for both groups. Furthermore, politically connected firms are sizeable, more leveraged, and have higher institutional ownership but less managerial ownership, per panel C of Table 4.

(Table 4 here)

5.3.2. Ethnicity

We extend the univariate analysis by separating the sample based on the median values of the ethnic groups: $BODFRANCO_{it}$ and $BODINDO_{it}$, as tabulated in Tables 5 and 6, respectively. Firms that have more than the median value of $BODFRANCO_{it}$ recorded significantly higher $LnPRICE_{it}$, higher earnings per share ($LnEPS_{it}$) and book value ($LnBV_{it}$), and fewer firms considered as politically connected ($POLCON_{it}$). Not surprisingly, firms above the median value of $BODFRANCO_{it}$ have significantly fewer $BODINDO_{it}$ directors. Furthermore, these firms are significantly smaller, have higher managerial ownership ($MANOWN_{it}$, Mann-Whitney only), and are significantly older than firms with less than the median value of BODFRANCO directors.

(Table 5 here)

We perform a similar test for *BODINDO_{it}*. We find no significant differences between the two groups of firms in *LnPRICE*, *LnEPS*, and *LnBV*. As expected, firms with more than the median value of *BODINDO_{it}* are likely to be connected. Further, firms with a higher *BODINDO_{it}* have many significantly lower *BODFRANCO_{it}* directors. Panel C of Table 6 presents the control variables' mean and median differences. Firms with more

than the median value of $BODINDO_{it}$ are significantly larger, have a significantly lower level of managerial ownership, and are younger firms.

These univariate analyses provide insight into the differences in firm characteristics for firms dominated by *BODFRANCO_{it}* or *BODINDO_{it}*. A significant finding is that firms that *BODINDO_{it}* dominates are more politically connected. In addition, the univariate analysis supports our argument that firms with a high level of *BODINDO_{it}* directors are also politically connected.

(Table 6 here)

5.4. Multivariate

Table 7 presents the main regressions. Column 1 of Table 7 presents the baseline regression for Ohlson's model, and we find that $LnEPS_{it}$ and $LnBV_{it}$ are positively and significantly associated with $LnPRICE_{it}$, at the one percent level. Column 2 of Table 7 tabulates the regression to include the control variables. We find a positive and significant relationship between $POLCON_{it}$ and $PRICE_{it}$ ($\beta = 0.130$, t=1.976, p<0.1), suggesting that political connections positively affect share prices. We find a negative and significant relationship between $BODFRANCO_{it}$ and $PRICE_{it}$ ($\beta = 0.309$, t=-4.266, p<0.01) and no significant relationship for $BODINDO_{it}$. All the control variables are significantly associated with $PRICE_{it}$, except $LNSIZE_{it}$ and LEV_{it} .

Column 3 of Table 7 presents the regression to test the value relevance of political connections (*POLCON_{it}*). We find the coefficient for *POLCON*BV_{it}* is negative and significant ($\beta = -0.180$, t = -2.688, p < 0.05), and this signals that political connections reduce the value relevance of book value (BV). However, we could not find any evidence for *POLCON*EPS_{it}*. All control variables remain similar to column 2 of Table 7. The

results overall support the notion that political connections decrease earnings quality. In addition, the results are similar to other political connection studies (please see Gul, 2006; Fung *et al.*, 2015).

(Table 7 here)

5.5. Testing for endogeneity

Political connections have been proven to be endogenous by several studies (Abdul Wahab *et al.*, 2011; Domadenik *et al.*, 2016). Maddala (1991) argues that selection bias happens once observations are sorted into discrete groups non-randomly, leaving way for potential coefficient bias in ordinary least squares procedures. Indeed, political connection scores are not allocated randomly but based on the firm's relationship with the government over time.

A two-step procedure is suggested: the first stage involves identifying the endogenous independent variable and addressing the selection model by using a probit regression to compute the inverse Mills ratio ($IMILLS_{it}$), which will be, in the second phase, added to the main regression. The following first-stage model is applied to determine the inverse Mills ratio ($IMILLS_{it}$):

 $POLCON_{it} = \beta_0 + \beta_1 BIG4_{it} + \beta_2 LnEPS_{it} + \beta_3 LnBV_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 LNSIZE_{it} + \beta_7 LEV_{it} + \beta_8 MANOWN_{it} + \beta_9 INSTOWN_{it} + \beta_{10} LNAGE_{it} + \beta_{11}CG2004_{it} + \beta_{12}INDUSTRY_{it}$ (Equation 5)

This regression equation differs slightly from the main regression equation as an exclusion restriction – BIG4 has been added to the first stage of probit regression. The first added variable, BIG4 is a dummy variable that holds the value of 1 if the auditor is

one of the Big 4 international auditing firms; 0 otherwise. Many studies have incorporated the relationship between the choice of auditors and political connections. Guedhami *et al.* (2014) investigate the relationship between political connections and auditor choice and find that connected firms are more likely to appoint a Big 4 auditor, as the insiders in these firms are eager to improve accounting transparency and convince outside investors that they are not misappropriating resources. Liu *et al.* (2017) examined the relationship between political connected firms and found that political connected firms hire auditors of lower quality. Cheng *et al.* (2015) reached the same conclusion.

However, several recent studies have found that political connections adversely determine the likelihood of engaging quality auditors. Cheng *et al.* (2015), Habib *et al.* (2017), and Liu *et al.* (2017) argue that the likelihood of engaging quality auditors by politically connected firms is driven by managerial incentives to distort numbers, expropriate assets, and mask related-party transactions. These studies find results consistent with their arguments. Therefore, it is argued that a relationship between political connections and the choice of auditors exists.

Column 1 of Table 8 presents the probit regression when we include $BIG4_{it}$ as a determinant for *POLCON*. The coefficient for $BIG4_{it}$ and the *IMILLS_{it}* variable in columns 2 and 3 of Table 8 are insignificant, suggesting that the regressions do not suffer from selection bias. After controlling for selection bias, the results remain qualitatively similar to Table 7.

(Table 8 here)

5.6. Political connections, board diversity and value relevance

Table 9 presents the regression results when we test the impact of the ethnic groups on the value relevance of political connections. Column 2 of Table 9 shows the preliminary test for the value relevance for $BODFRANCO_{it}$ and $BODINDO_{it}$. We find $BODFRANCO_{it}$ and $BODINDO_{it}$ reduce the value relevance of EPS_{it} on $PRICE_{it}$. However, we find contrasting results for BV_{it} .

Column 3 of Table 9 tabulates the regression for *BODFRANCO*_{it}. The coefficient for *POLCON*BODFRANCO*EPS*_{it} is positive and significant (β =0.639, *t*=3.434, *p*<0.01). This provides evidence that *BODFRANCO*_{it} mitigates the negative impact of political connections on the value relevance of *EPS*_{it}. Column 4 of Table 9 presents the result for *BODINDO*_{it}. The interaction term *POLCON*BODINDO*EPS*_{it} is negative and significant (β =-1.251, t=-3.320, *p*<0.01). The results support our arguments that *BODFRANCO*_{it} directors are experienced and probably more trustworthy to various stakeholders and mitigate the negative perception of political connections.

Furthermore, we find that the negative impact of political connections on value relevance worsens with firms with a high level of *BODINDO*_{it} directors. The results are not surprising since *BODINDO*_{it} directors are often linked to the government, and their presence on the board could be largely ceremonial or an indicator of trying to tap into government grants. This indicator is usually perceived negatively since it signals the firm's inability to tap into the external market.

(Table 9 here)

6. Additional Analyses

We extend the test by separating the sample based on the median values of the ethnic

groups; BODFRANCO_{it} and BODINDO_{it}. The premise of our test is to investigate the collective impact of ethnic groups in mitigating the value relevance of political connections. The directors from various ethnic groups are expected to interact on the board. We argue that firms with Franco-Mauritian directors' domination will weaken the higher negative impact caused by the presence of Indo-Mauritian directors. In contrast, we argue that the positive impact of Franco-Mauritian directors will be weakened in firms that Indo-Mauritians dominate. To investigate these conjectures, we re-run the regressions presented in Table 9 by separating the samples according to the median values of BODFRANCO_{it} and BODINDO_{it}, with minor adjustments. Table 10 presents the results for the impact of *BODFRANCO*_{it} and *BODINDO*_{it} directors, respectively. Columns 1 and 2 present the regressions when we separate the sample based on the median values of *BODFRANCO*_{it}. The results are similar to those tabulated in Tables

and 9. We find the interactions POLCON*BODINDO*LnEPS_{it} and *POLCON*BODINDO*LnBV_{it}* are not significant.

Columns 3 and 4 of Table 10 tabulate the regression when we split the sample based on the median value of BODINDO_{it}. We find that the results for the interactions POLCON*BODFRANCO*EPS_{it} and POLCON*BODFRANCO*BV_{it} are not significant, suggesting the impact of the presence of BODINDO_{it} is relatively minimal or non-NO. existence.

(Table 10 here)

6.1. Elections

During the sample period, Mauritius experienced three elections in 2005, 2010, and 2014. We assigned dummy variables to control for each of the elections. The untabulated results find that the election years are significantly related to *PRICE*. The main test variables' results remain qualitatively similar to Table 7.

7. Conclusion

 This study has two central objectives. The first and baseline objective is establishing the relationship between Mauritius's political connections and value relevance. We are motivated to conduct such an examination as there is limited evidence of Mauritius's political connections. The political landscape in Mauritius is very interesting and is seen as a country dominated by two prominent families. Based on our analysis of 541 firm-year observations for 2001-2016 and the operationalization of political connections based on Faccio's (2006) seminal work, we find that political connections mitigate the value relevance of earnings and book value. The results are consistent with studies (e.g. Chen *et al.*, 2010; Chaney *et al.*, 2011) that examine the impact of political connections on earnings quality. Further, the results remain similar after we perform a selection test.

The second and primary research objective is to examine whether board ethnicity mitigates the negative impact of political connections on value relevance. Mauritius is blessed with diverse ethnic groups working collectively since gaining independence. We conjecture that the ethnic groups in Mauritius represent the political landscape. We chose two contrasting ethnic groups: Franco-Mauritians, who dominate the capital market but are a minority in the population, and Indo-Mauritians, who control the government and are a majority.

We find contrasting results based on the two ethnic groups, Franco-Mauritians, and Indo-Mauritians. We find firms that have Franco-Mauritians as directors mitigate the negative impact of political connections on the value relevance of earnings and book value. In contrast, we find that Indo-Mauritian directors worsen the negative impact of political connections on value relevance. The contrasting findings highlight the unique

institutional settings in Mauritius. The results add to the extant literature, especially on the role of ethnicity in the involvement of the capital market via political connections. The findings of this study should act as a catalyst for further research on political connections in this region, which is often deemed 'sensitive' for accounting researchers

³ Mauritius has an anti-corruption score of 73.1 out of 100 in the Mo Ibrahim Index in 2020 and ranked second out of 54 in Africa (Mo Ibrahim Foundation, 2021

⁴ It is a customized version of the Westminster majoritarian model whereby the constitution appoints for a Head of State or President, Vice-President, Prime Minister or Head of Government, Deputy Prime Minister(s), Members of Parliament, Parliamentary opposition with an appointed Leader of the opposition and many parliamentary secretaries.

⁵ The industries are Banks and Insurance; Commerce; Manufacturing; Investments; Leisure and Hotels; Property Development; Sugar Industry and Transport.

¹ The Court found him not guilty in November 2019 as the pieces of evidence presented by the prosecuting parties were vague and uncertain and therefore failed the test of certainty required under the provisions of the Law (Police V N. Ramgoolam, 2019).

² Mauritius had a Franco-Mauritian Prime Minister; Paul Berenger, between 2003-2005.

References

- Aapravasi Ghat Trust Fund (2015), "*Factsheet Aapravasi Ghat World Heritage Site*", available at: https://aapravasi.govmu.org/aapravasi/wpcontent/uploads/2020/09/Factsheet-Aapravasi-Ghat-World-Heritage-Site.pdf (accessed 27 September 2021)
- Abdul Wahab, E.A., Zain, M.M. and James, K. (2011), "Political connections, corporate governance and audit fees in Malaysia", Managerial Auditing Journal, Vol. 26 No. 5, pp. 393–418. <u>https://doi.org/10.1108/02686901111129562</u>
- Abdul Wahab, E.A., Pitchay, A.A. and Ali, R. (2015), "Culture, corporate governance and analysts forecast in Malaysia", Asian Review of Accounting, Vol 23 No. 3, pp. 232-255. <u>https://doi.org/10.1108/ARA-03-2014-0033</u>
- Abdul Wahab, E.A., Madah Marzuki, M., Jaafar, S.B. and Masron, T.A. (2018), "Board diversity and total directors' remuneration: evidence from an emerging market", Pacific Accounting Review, Vol. 30 No. 2, pp. 243–272. https://doi.org/10.1108/PAR-02-2016-0021
- Abdullah, S.N. (2013), "Causes of gender diversity in Malaysian firms", Journal of Management and Governance, Vol. 18 No. 4, pp. 1137-1159. https://doi.org/10.1007/s10997-013-9279-0
- Adedeji, S.B. and Kajola, S.O. (1998), Cost Accounting Concept, Analysis and Use for Decision Making. Centre for Sandwich Programme (CESAP).
- Adhikari, A., Derashid, C. and Zhang, H. (2006), "Public Policy, Political Connections, and Effective Tax Rates: Longitudinal Evidence from Malaysia", Journal of Accounting and Public Policy, Vol. 25, pp. 574–95. https://doi.org/10.1016/j.jaccpubpol.2006.07.001
- Agarwal, A. and Knoeber, C.R. (2001), "Do some outside directors play a political role?", Journal of Law and Economics, Vol. 44 No. 1, pp. 179–198. https://doi.org/10.1086/320271
- Arfken, D.E., Bellar, S.L. and Helms, M. M. (2004), "The ultimate glass ceiling revisited: The presence of women on corporate boards", Journal of Business Ethics, Vol. 50 No. 2, pp. 177–186. https://doi.org/10.1023/B:BUSI.0000022125.95758.98
- Badu, B. and Appiah, K.O. (2018), "Value relevance of accounting information: an emerging country perspective", Journal of Accounting and Organizational Change, Vol. 14 No. 4, pp. 473–491. <u>https://doi.org/10.1108/JAOC-07-2017-0064</u>
- Ball, R. and Brown, P. (1968), "An Empirical Evaluation of Accounting Income Numbers", Journal of Accounting Research, Vol. 6 No. 2, pp. 159–178. https://doi.org/10.2307/2490232
- Ball, R., Robin, A. and Wu, J. (2003), "Incentives versus standards: Properties of

 accounting income in four East Asian countries", Journal of Accounting and Economics, Vol. 36, pp. 235–270. https://doi.org/10.1016/j.jacceco.2003.10.003

- Baranchuk, N. and Dybvig, P. (2009), "Consensus in diverse corporate boards", Review of Financial Studies, Vol. 22, pp. 715–747. https://doi.org/10.1093/rfs/hhn052
- Bart, C. and McQueen, G. (2013), "Why women make better directors?", International Journal of Business Governance and Ethics, Vol. 8, pp. 93–99.
- Barth, M.E., Beaver, W.H. and Landsman, W.R. (2001), "The Relevance of the Value Relevance Literature For Financial Accounting Standard Setting: another view", Journal of Accounting and Economics, Vol. 31 No. 1, pp. 77–104. https://doi.org/10.1016/S0165-4101(01)00019-2
- Batta, G., Sucre Heredia, R. and Weidenmier, M. (2014), "Political Connections and Accounting Quality under High Expropriation Risk", European Accounting Review, Vol. 23 No. 4, pp. 485–517. https://doi.org/10.1080/09638180.2014.906316
- Beaver, W.H. (1968), "The Information Content of Annual Announcements Earnings", Journal of Accounting Research, Vol. 6, pp. 67–92. https://doi.org/10.2307/2490070
- Beaver, W.H. (2002), "Perspectives on recent capital market research", Accounting Review, Vol. 77 No. 2, pp. 453–474. https://doi.org/10.2308/accr.2002.77.2.453
- Beaver, W.H., Lambert, R. and Morse, D. (1980), "The information content of security prices. Journal of Accounting and Economics", Vol. 2 No. 1, pp. 3-28. https://doi.org/10.1016/0165-4101(80)90013-0
- Beebeejaun, A. and Koobloll, J. (2018), "An assessment whether shareholder activism can be a corporate governance driver in the case of Mauritius", International Journal of Law and Management, Vol. 60 No. 6, pp. 1313–1337. https://doi.org/10.1108/IJLMA-07-2017-0154
- Beisland, L., Hamberg, A.M. and Novak, J. (2010), "The Value Relevance Across Industries: What Happened to the New Economy?", Financial Management Association. http://www.fma.org/Prague/Papers/Value Relevance across Industries.pdf.
- Bertrand, M., Kramarz, F., Schoar, A. and Thesmar, D. (2004), "Politically Connected CEOs and Corporate Outcomes: Evidence from France", working paper, University of Chicago.
- Bilimoria, D. and Piderit, S. (1994), "Board committee membership: effects of sex-based bias", Academy of Management Journal, Vol. 37, pp. 1453–1477. https://doi.org/10.2307/256795
- Boonlert-U-Thai, K. and Schaberl, P. (2022), "Value relevance of book values, earnings, and future earnings: Evidence by time, life cycle stage, and market uncertainty". Asian Review of Accounting, ahead-of-print.

- Boubakri, N., Guedhami, O., Mishra, D. and Saffar, W. (2012), "Political connections and the cost of equity capital", Journal of Corporate Finance, Vol. 18 No 3, pp. 541–559. https://doi.org/10.1016/j.jcorpfin.2012.02.005
- Brautigam, D. and Diolle, T. (2009), "Coalitions, capitalists, and credibility: Overcoming the crisis of confidence at independence in Mauritius", research paper 04, Developmental Leadership Program. available at: https://www.dlprog.org/publications/research-papers/coalitions-capitalists-andcredibility-overcoming-the-crisis-of-confidence-at-independence-in-mauritius (accessed 23 March 2018)
- Bunwaree, S. and Kasenally, R. (2005), "Political Parties and Democracy in Mauritius", Research Report No 19, EISA, South Africa.
- Bushman, R., Chen, Q., Engel, E. and Smith, A. (2004), "Financial accounting information, organizational complexity and corporate governance systems", Journal of Accounting and Economics, Vol. 37, pp. 167-201. https://doi.org/10.1016/j.jacceco.2003.09.005
- Carroll, B.W. and Carroll, T. (2000), "Trouble in paradise: Ethnic conflict in Mauritius", Journal of Commonwealth & Comparative Politics, Vol. 38 No, 2, pp. 25–50. https://doi.org/10.1080/14662040008447817
- Carter, D.A., Souza, F.D., Simkins, B.J. and Simpson, W.G. (2010), "The Gender and Ethnic Diversity of US Boards and Board Committees and Firm", Corporate Governnace: An International Review, Vol. 18 No. 5, pp. 396–414. https://doi.org/10.1111/j.1467-8683.2010.00809.x
- Chaney, P.K., Faccio, M. and Parsley, D. (2011), "The quality of accounting information in politically connected firms", Journal of Accounting and Economics, Vol. 51, pp. 58–76. <u>https://doi.org/10.1016/j.jacceco.2010.07.003</u>
- Chen, C.J.P., Ding, Y. and Kim, C. (2010), "High-Level Politically Connected Firms, Corruption, and Analyst Forecast Accuracy Around the World", Journal of International Business Studies, Vol. 41, pp. 1505 - 1524. https://doi.org/10.1057/jibs.2010.27
- Cheng, C.L., Hsu, C.S. and Kung, F.H. (2015), "Political connections, managerial incentives and auditor choice: evidence from China", Pacific Accounting Review, Vol. 27 No. 4, pp. 441–465. https://doi.org/10.1108/PAR-10-2013-0096
- Chiriyankandath, J. (2009), "Global Indian diasporas: Exploring trajectories of migration and theory", Journal of Global History, Vol. 4 No. 3, pp. 500-502. https://doi.org/10.1017/S1740022809990209
- CIA (2021), "The world factbook", available at: https://www.cia.gov/the-world-factbook/countries/mauritius/#people-and-society (accessed 6 December 2021)
- Civilize, S., Wongchoti, U. and Young, M. (2015), "Political Connection and Stock Returns: A Longitudinal Study", The Financial Review, Vol. 50 No. 1, pp. 89– 119. https://doi.org/10.1111/fire.12061

- Collins, D.W., Maydew, E.L. and Weiss, I.S. (1997), "Changes in the value-relevance of earnings and book values over the past forty years", Journal of Accounting and Economics, Vol. 24 No. 1, pp. 39–67. <u>https://doi.org/10.1016/S0165-4101(97)00015-3</u>
- Dale-Olsen, H., Schone, P. and Verner, M. (2013), "Diversity among Norwegian boardsnof directors: Does a quota for women improve performance", Feminist Economics, Vol. 19 No. 4, pp. 110-135. https://doi.org/10.1080/13545701.2013.830188
- Darga, L. A. and Joomun, G.D. (2005), "Mauritius", South African Journal of International Affairs, Vol. 12 No. 1, pp. 95–110. https://doi.org/10.1080/10220460509556751
- Diftar, D. and Elkalla, T. (2019), "*The value relevance of accounting information in the MENA region*", Journal of Financial Reporting and Accounting, Vol. 17 No. 3, pp. 519-539. https://doi.org/10.1108/JFRA-09-2018-0079
- Dinç, I.S. (2005), "Politicians and banks: Political influences on government-owned banks in emerging markets", Journal of Financial Economics, Vol. 77 No. 2, pp. 453–479. https://doi.org/10.1016/j.jfineco.2004.06.011
- Distefano, J.J. and Maznevski, M.L. (2000), "Creating value with diverse teams in global management", Organizational Dynamics, Vol. 29 No. 1, pp. 45–63. https://doi.org/10.1016/S0090-2616(00)00012-7
- Domadenik, P., Prašnikar, J. and Svejnar, J. (2016), "Political Connectedness, Corporate Governance, and Firm Performance", Journal of Business Ethics, Vol. 139 No. 2, pp. 411–428. https://doi.org/10.1007/s10551-015-2675-4
- Faccio, M. (2006), "Politically connected firms", The American Economic Review, Vol. 96 No. 1, pp. 369–386. https://doi.org/10.1257/000282806776157704
- Faccio, M., Masulis, R.W. and McConnell, J.J. (2006), "Political Contributions and Corporate Bailouts", The Journal of Finance, Vol. 61 No. 6, pp. 2597–2635. https://doi.org/10.1111/j.1540-6261.2006.01000.x
- Fan, J.P.H., Wong, T.J. adnd Zhang, T. (2007), "Politically connected CEOs, corporate governance, and Post-IPO performance of China's newly partially privatized firms", Journal of Financial Economics, Vol. 84 No. 2, pp. 330–357. https://doi.org/10.1016/j.jfineco.2006.03.008
- Fisman, R. (2001), "Estimating the Value of Political Connections", The American Economic Review, Vol. 91 No. 4, pp. 1095–1102. https://doi.org/10.1257/aer.91.4.1095
- Francis, J. and Schipper, K. (1999), "Have Financial Statements Lost Their Relevance?", Journal of Accounting Research, Vol. 37 No. 2, pp. 319–352. https://doi.org/10.2307/2491412
- Fung, Y.K., Gul, F.A. and S. Radhakrishnan. (2015), "Corporate politically connections and the 2008 Malaysian election", Accounting Organization and Society, Vol. 43
No. 2, pp. 67-86. https://doi.org/10.1016/j.aos.2015.04.001

- Gist, W.E. and Abdul Wahab, E.A. (2021), "Political Patronage, Audit Quality and the Properties of Analysts' Earnings Forecasts in Malaysia", Journal of International Accounting Research, Vol. 20 No 1, pp. 49–78. https://doi.org/10.2308/JIAR-19-
- Glasscock, R., Korenok, O. and Dorminey, J. (2022), "Spurious Correlation Due to Scaling", Journal of Accounting, Auditing & Finance, https://doi.org/10.1177/0148558X2110637040148558X211063704.
- Goodstein, J., Gautam, K. and Boeker, W. (1994), "The effects of board size and diversity on strategic change", Strategic Management Journal, Vol. 15 No. 3, pp. 241-250. https://doi.org/10.1002/smj.4250150305
- Guedhami, O., Pittman, J.A. and Saffar, W. (2014), "Auditor Choice in Politically Connected Firms", Journal of Accounting Research, Vol. 52 No. 1, pp. 107-162. https://doi.org/10.1111/1475-679X.12032
- Gul, F.A. (2006), "Auditors' response to political connection and cronyism in Malaysia", Journal of Accounting Research, Vol. 44 No. 5, pp. 931-963. https://doi.org/10.1111/j.1475-679X.2006.00220.x
- Habib, A., Muhammadi, A.H. and Jiang, H. (2017), "Political Connections and Related Party Transactions: Evidence from Indonesia", International Journal of Accounting, Vol. 52 No. 1, pp. 45–63. https://doi.org/10.1016/j.intacc.2017.01.004
- Hanousek, J., Kocenda, E. and Svejnar, J. (2007), "Origin and concentration: Corporate ownership, control and perfonance in firms after privatization", Economics of Transition, Vol. 15 No. 1, pp. 1-31. https://doi.org/10.1111/j.1468-0351.2007.00278.x
- Hillman, A.J., Shropshire, C. and Cannella, A.A. (2007), "Organizational predictors of women on corporate boards", Academy of Management Journal, Vol. 50 No. 4, pp. 941–952. htts://doi.org/ 10.5465/AMJ.2007.26279222
- Holmberg, L. (1962). "Mauritius. A study in disaster", Economy and History, Vol. 5 No. 1, pp. 3-29. https://doi.org/10.1080/00708852.1962.10418985
- Johnston, D. and Malina, M.A. (2008), "Managing sexual orientation diversity: The impact on firm value", Group and Organisation Management, Vol. 33, pp. 602–625. <u>https://doi.org/10.1177/1059601108321833</u>
- Johnson, S. and Mitton, T. (2003), "Cronyism, and capital controls: Evidence from Malaysia", Journal of Financial Economics, Vol. 67, pp. 351-382. https://doi.org/10.1016/S0304-405X(02)00255-6
- Jones, S.L., Megginson, W.L., Nash, R.C. and Netter, J.M. (1999), "Share issue privatizations as financial means to political and economic ends", Journal of Financial Economics, Vol. 53 No. 2, pp. 217-253. <u>https://doi.org/10.1016/S0304-405X(99)00021-5</u>

- Khwaja, A.I. and Mian, A. (2005), "Do Lenders Favor Politically Connected Firms? Rent Provision in an Emerging Financial Market", The Quarterly Journal of Economics, Vol. 120 No. 4, pp. 1371–1411. https://doi.org/10.1162/003355305775097524
 - Kim, M. and Kross, W. (2005), "The ability of earnings to predict future operating cash flows has been increasing - Not decreasing", Journal of Accounting Research, Vol. 43 No. 5, pp. 753–780. https://doi.org/10.1111/j.1475-679X.2005.00189.x
 - Kim, Y., Li, S., Pan, C. and Zuo, L. (2013), "The role of accounting conservatism in the equity market: evidence from seasoned equity offerings", The Accounting Review, Vol. 88 No. 4, pp. 1327–1356. https://doi.org/ 10.2308/accr-50420
 - Kothari, S. (2001), "Capital markets research in accounting", Journal of Accounting and Economics, Vol. 31 No. 1–3, pp. 105–231. <u>https://doi.org/10.1016/S0165-4101(01)00030-1</u>
 - Lange, M. (2003), "Embedding the colonial state: A comparative-historical analysis of state building and broad-based development in Mauritius", Social Science History, Vol. 27, pp. 397-423. https://doi.org/10.1017/S014555320001258X
 - Leuz, C. and Oberholzer-Gee, F. (2006), "Political relationships, global financing, and corporate transparency: Evidence from Indonesia", Journal of Financial Economics, Vol. 81 No. 2, pp. 411–439. https://doi.org/10.1016/j.jfineco.2005.06.006
 - Li, H., Meng, L., Wang, Q. and Zhou, L. A. (2008), "Political connections, financing and firm performance: Evidence from Chinese private firms", Journal of Development Economics, Vol. 87 No. 2, pp. 283–299. https://doi.org/10.1016/j.jdeveco.2007.03.001
 - Liang, L.W., Lin, T.J., Yu, H.F. and Li, Y. W. (2021), "The Impact of Political Connection and Board Diversity on Company Performance: Evidence from China", Managerial and Decision Economics, Vol. 43 No. 6, pp. 2347-2357. https://doi-org.ezproxy.mdx.ac.uk/10.1002/mde.3529
 - Lincoln, A. and Adedoyin, O. (2012), "Corporate governance and gender diversity in Nigerian boardrooms", World Academy of Science, Engineering and Technology, Vol. 71, pp. 1853–1859. https://ssrn.com/abstract=2201160
 - Lipton, M. and Lorsch, J. (1992), "A modest proposal for improved corporate governance", Business Lawyer, Vol. 48, pp. 59–77. https://heinonline.org/HOL/P?h=hein.journals/busl48&i=95.
 - Liu, Y., Li, X., Zeng, H. and An, Y. (2017), "Political connections, auditor choice and corporate accounting transparency: evidence from private sector firms in China", Accounting and Finance, Vol. 57 No. 4, pp. 1071–1099. https://doi.org/10.1111/acfi.12207
 - Liu, J. and Liu, C. (2007), "Value relevance of accounting information in different stock market segments: The case of Chinese A-, B- and H-shares", Journal of International Accounting Research, Vol. 6, pp. 55–81.

https://doi.org/10.2308/jiar.2007.6.2.55

- Lubberink, M.J.P., and Willett, R.J. (2021), "Specification of the Market-Accounting Relation", Available at SSRN 3976635.
- Luckerath-Rovers, M. (2013), "Women on boards and firm performance", Journal of Management and Governance, Vol. 17, pp. 491–508. https://doi.org/10.1007/s10997-011-9186-1
- Maddala, G. (1991), "A perspective on the use of limited-dependent and qualitative variables models in accounting research", The Accounting Review, Vol. 66, No. 4, pp. 788–807.
- Mahadeo, J.D., Soobaroyen, T. and Hanuman, V.O. (2012), "Board Composition and Financial Performance: Uncovering the Effects of Diversity in an Emerging Economy", Journal of Business Ethics, Vol. 105 No. 3, pp. 375–388. https://doi.org/10.1007/s10551-011-0973-z
- Mahadeo, J.D. and Soobaroyen, T. (2013), "A Longitudinal Study of the Implementation of the Corporate Governance Code in a Developing Country: The Case of Mauritius", Business and Society, Vol. 55 No. 5, pp. 738–777. https://doi.org/10.1177/0007650313501838
- Mishra, R.K. and Jhunjhunwala, S. (2013), "Diversity—Toward an Effective Board", R.
 K. Mishra & S. Jhunjhunwala (Eds.), *Diversity and the Effective Corporate Board*, Academic Press, Cambridge, Massachussetts, pp. 1–11
- Munisi, G. and Randøy, T. (2013), "Corporate governance and company performance across Sub-Saharan African countries", Journal of Economics and Business, Vol. 70, pp. 92–110. https://doi.org/10.1016/j.jeconbus.2013.08.003
- Nee, V., Opper, S. and Wong, S. (2007), "Development state and corporate governance in China", Management and Organization Review, Vol. 3 No. 1, pp. 19-53. https://doi.org/10.1111/j.1740-8784.2007.00061.x
- Negakis, J. (2005), "Accounting and capital markets research: A review", Managerial Finance, Vol. 31 No. 2, pp. 1-23.
- Ntim, C.G. (2013), "Board diversity and organizational valuation: unravelling the effects of ethnicity and gender", Journal of Management and Governance, Vol. 19 No. 1, pp. 167–195. https://doi.org/10.1007/s10997-013-9283-4
- Ohlson, J.A. (1995), "Earnings, book-values, and dividends in equity valuation", Contemporary Accounting Research, Vol. 11 No. 11, pp. 661–687. https://doi.org/10.1111/j.1911-3846.1995.tb00461.x
- POLICE v N. RAMGOOLAM. (2019). The Intermediate Court of Mauritius.
- Reuters (2016), "Mauritius environment minister resigns over graft allegations". Available at: https://www.reuters.com/article/ozatp-uk-mauritius-corruptionidAFKCN0WP1LZ

י ר	
2	
3	
4	
5	
6	
7	
8	
9	
10	
10	
11	
12	
13	
14	
15	
16	
17	
18	
10	
19	
20	
21	
22	
23	
24	
25	
26	
27	
27	
28	
29	
30	
31	
32	
33	
34	
35	
26	
20	
3/	
38	
39	
40	
41	
42	
43	
44	
15	
45	
40	
4/	
48	
49	
50	
51	
52	
52	
22	
54	
55	
56	
57	
58	

59

60

Riahi-Belkaoui, A. (2004), "Politically-Connected Firms: Are They Connected To Earnings Opacity?", Research in Accounting Regulation, Vol. 17 No 1, pp. 25– 38. <u>https://doi.org/10.1016/S1052-0457(04)17002-1</u>

- Roberson, Q.M. and Park, H.J. (2007), "Examining the link between diversity and firm performance: The effects of diversity reputation and leader racial diversity", Group and Organization Management, Vol. 32, pp. 548–568. https://doi.org/10.1177/1059601106291124
- Salverda, T. (2013), "Balancing (re)distribution: Franco- Mauritians Landownership in the Maintenance of an Elite Position", Journal of Contemporary African Studies, Vol. 31 No. 3, pp. 503-521. https://doi.org/10.1080/02589001.2013.812455
- Salverda, T. (2015), "The Franco-Mauritian Elite: Power and Anxiety in the Face of Change", Berghahn Books, Incorporated.
- Simkins, B.J. amd Carter, D.A. (2003), "Corporate Governance, Board Diversity, and Firm Value", The Financial Review, Vol. 38, pp. 33–53. https://doi.org/10.1111/1540-6288.00034
- Sonnenfeld, J.A. (2002), "What makes great boards great", Harvard Business Review, Vol. 80 No. 9, pp. 106-113.
- Srebrnik, H. (2002), "'Full of sound and fury': Three decades of parliamentary politics in Mauritius. Journal of Southern African Studies", Vol. 28 No. 2, pp. 277–290. https://doi.org/10.1080/03057070220140702
- Srinidhi, B., Gul, F.A. and Tsui. J. (2011), "Female directors and earnings quality", Contemporary Accounting Research, Vol. 28 No. 5, pp. 1610–1644. https://doi.org/10.1111/j.1911-3846.2011.01071.x
- Tantawy, S.M. and Moussa, T. (2023), "The effect of political connections on firms' auditor choice decisions and audit opinions: evidence from Egypt", Asian Review of Accounting, ahead-of-print. https://doi-org.ezproxy.mdx.ac.uk/10.1108/ARA-07-2022-0161
- Tee, C.M. (2019), "Political connections, the cost of debt and board attributes: evidence from Malaysia", Managerial Finance, Vol. 45 No. 7, pp. 842–855. https://doi.org/10.1108/MF-04-2018-0179
- Triana, M.C., Miller, T.L. and Trezebiatowski, T.M. (2014), "The Double-Edged Nature of Board Gender Diversity: Diversity, Firm Performance, and the Power of Women Directors as Predictors of Strategic Change", Organization Science, Vol. 5 No. 2, pp. 609–32. https://doi.org/10.1287/orsc.2013.0842
- Van Der Walt, N. and Ingley, C. (2003), "Board dynamics and the influence of professional background, gender and ethnic diversity of directors", Corporate Governance: An International Review, Vol. 11, pp. 218–234. https://doi.org/10.1111/j.1467-8683.2005.00456.x
- Wan, J. (2015), "Changing fortunes of Mauritius' ousted leaders", *Al Jazeera*, 17 March. Available at:

https://www.aljazeera.com/features/2015/3/17/changing-fortunes-of-mauritiusousted-leaders

- <text>

APPENDIX A

Operational Definitions

Variable	Operational Definition	Source(s)
Panel A: Depen	dent Variable	
LnPRICE	Log transformation of Share Price of firm <i>i</i>	Stock Exchange of
	for the period <i>t</i> in Mauritian Rupees.	Mauritius
Panel B: Indepe	ndent Variables	
LnEPS	Natural log transformation of Earnings per	DataStream, BvD's
	share of firm <i>i</i> for the period <i>t</i> in Mauritian	Osiris, annual reports
	Rupees.	
LnBV	Natural log transformation of Book value per	DataStream, BvD's
	share of equity of firm <i>i</i> at time <i>t</i> in Mauritian	Osiris, annual reports
	Rupees.	
POLCON	Political connection of firm <i>i</i> at time <i>t</i> .	
BODFRANCO	Franco-Mauritian directors on the board of	Annual reports
	directors of firm <i>i</i> .	
BODINDO	Indo-Mauritian directors from Indian descent on the board of directors of firm <i>i</i>	Annual reports
Panel C: Contro	l Variables	
SIZE	Total Assets of firm <i>i</i> at time <i>t</i> in millions of	DataStream, BvD's
	Mauritian Rupees.	Osiris, annual reports
LNSIZE	Natural log transformation of Total Assets of	DataStream, BvD's
	firm <i>i</i> at time <i>t</i> in millions of Mauritian	Osiris, annual reports
	Rupees.	
LEV	Natural log transformation of Total Debt	DataStream, BvD's
MANOWN	divided by 1 otal Assets.	Osiris, annual reports
MANOWN	Percentage of direct managerial ownership of	DataStream, BVD's
NICTOWNI	IIIM <i>l</i> .	Osiris, annual reports
INSIOWN	institutional shoreholders of firm i	DataStream, BVD'S
ACE	Institutional shareholders of firm <i>l</i> .	A noused reports
AUL	Ivatural log transformation of the sum of	Annual reports
	years since incorporation	
Panel D: Exclus	ion Restriction	
BIG4	An indicator variable that takes the value of 1	Annual reports
	if the firm engaged an international Big 4	1
	auditor	

APPENDIX B

Politically connected firms

Politically connected firms were identified using Faccio (2006) 's definition of politically connected firms and/or significant government ownership in the firm. Therefore, politically connected firms satisfy any of the following:

- (a) Had Member of Parliaments (MPs) on the board of directors,
- (b) Had senior executives closely related to a MP or government official,
- (c) Had significant ownership by government or government-linked organizations.

The data was collected through:

- (a) Connections identified by linking the biodata of MPs and Government officials from the government websites and the biodata of senior executives in annual reports,
- (b) Percentage of ownership of government and government-related entities in annual report and companies' websites
- (c) Review of newspapers, government publications and firms' publications to established connections
- (d) Review of biodata of board members in annual reports.

Due to the sensitive nature of political connections in Mauritius and the current political turmoil in the country, the names of the firms are not disclosed:

Firm A	1.	A board member was the lawyer of one of the Prime ministers of
		Mauritius. The connection was identified through the obituary of this
		person as he passed away in 2013.
	2.	A board member was the brother of an MP: The connection was
		established through the biodata of the MP on the government website and
		the biodata of the director.
	3.	A board member was the sister-in-law of the Minister of Education and
		MP, she was the CEO of a state-owned organization and director on
		several government boards. This connection was established through the
		newspapers, the biodata of the Minister on the government website and
		the websites of the different organizations, of which she is a director.
	4.	Two government entities were also considered as major shareholders for
		the period 2009 – 2016. Connections identified in the annual reports.

Table 1: Sample Selection

Description	Sample Size
Firm-year Observations 2001-2016 ^a	688
Less Observations from Financial Companies	(48)
	640
Less Observations with missing financial data	(99)
Firm-year Observations for the final sample	541

rig with all audit, ing the 2001-2016 pc. Stock Exchange of Maun. ^a The sample was developed starting with all audited firms listed on the Stock Exchange of Mauritius during the period. During the 2001-2016 period of this study, there were 43 firms listed on the main market of the Stock Exchange of Mauritius.

	Mean	Median	Maximum	Minimum	SD
Panel A: Dependent variable	le				
PRICE	62.371	40.275	515	0.000	71.571
LnPRICE	3.526	3.696	6.244	-0.236	1.250
Panel B: Independent varia	bles				
EPS	260.058	3.295	146372.5	-39.77	6109.267
LnEPS	1.111	1.401	11.894	-6.645	1.632
BV	7.135	10.000	10.700	0.010	3.945
LnBV	1.544	2.303	2.370	-4.605	1.272
POLCON	0.366	0.000	1.000	0.000	0.482
BODFRANCO	0.607	0.667	1.000	0.000	0.314
BODINDO	0.174	0.100	1.000	0.000	0.207
Panel C: Control variables					
SIZE (millions)	13068.601	3257.542	317704.8	38.895	31859.363
LNSIZE	21.032	21.502	26.484	11.905	2.847
LEV	-1.0945	-0.676	0.0284	-6.308	1.243
MANOWN	0.0280	0.0002	7.361568	0.000	0.319
INSTOWN	0.513	0.596	0.910	0.000	0.234
AGE	62.84	41	227	1.000	52.38
LNAGE	3.841	3.714	5.425	0.000	0.783

Table 2: Descriptive Statistics

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. The logged variables LnPRICE, LnEPS, LnBV, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation.

Table 3: Correlation Matrix

Correlations		1	2	2	4	5	6	7	8	9	10	11
PRICE	1		.675***	.487***	.014	.090**	085**	.171***	.313***	.091**	.123***	.475***
EPS	2	.549***		.363***	.026	.006	035	.166***	.270***	007	004	.337***
BV	3	.289***	.200***		.031	.111**	051	046	.217***	.082*	.075*	.131***
POLCON	4	.058	.046	.079*		245***	.309***	.122***	.144***	.134***	109**	027
BODFRANCO	5	.055	.015	.077*	239***		727***	143***	194***	006	.175***	.194***
BODINDO	6	130***	053	010	.395***	649***		.183***	.130***	026	252***	113***
LNSIZE	7	.014	.112***	121***	.101**	137***	.196***		.402***	073*	.096**	.111**
LEV	8	.240***	.088**	.258***	.053	.016	139***	.121***		.248***	.148***	.319***
INSTOWN	9	.086**	069	.107**	.148***	018	048	174***	.346***		207***	.080**
MANOWN	10	.065	.018	.038	062	.004	047	.032	.039	022		019
LNAGE	11	.397***	.266***	.165***	009	.261***	265***	013	.342***	.033	050	

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the ,fio... sum of years since incorporation.

Significance at: *10, **5 and ***1 per cent levels.

4
5
6
7
8
9
10
11
11
12
13
14
15
16
17
18
19
20
20 21
21
22
23
24
25
26
27
28
29
20
20
31
32
33
34
35
36
37
38
39
10
40 41
41
42
43
44
45
46
47
48
49
50
51
51
52
53
54
55
56
57
58

Table 4: Univariate Analysis of Differences variables between Politically and Non-Politically
connected Firms in Mauritius (2001-2016)

	(n=	=199)	(n=	=342)	t - Tests	Mann	
	POL	POLCON =1		CON =0		Whitney	
	Mean	Median	Mean	Median	p-Value	p-Value	
Panel A: Dependent vari	able						
PRICE	67.121	37.700	58.589	40.650	0.183	0.747	
LnPRICE	3.592	3.630	3.444	3.705	0.187	0.747	
Panel B: Independent va	riables						
EPS	5.401	3.330	4.492	2.975	0.300	0.588	
LnEPS	1.252	1.401	0.965	1.340	0.054	0.171	
BV	7.592	10.000	6.918	10.000	0.054	0.484	
LnBV	1.584	2.303	1.540	2.303	0.694	0.484	
BODFRANCO	0.512	0.556	0.665	0.750	0.000	0.000	
BODINDO	0.278	0.222	0.112	0.091	0.000	0.000	
Panel C: Control variable	les						
LNSIZE	21.418	21.942	20.814	21.455	0.017	0.004	
LEV	-0.999	-0.553	-1.137	-0.718	0.210	0.001	
MANOWN	0.002	0.000	0.043	0.000	0.151	0.010	
INSTOWN	0.561	0.602	0.490	0.562	0.001	0.002	
LNAGE	3.845	3.689	3.860	3.738	0.828	0.526	

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. The logged variables *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is the natural log transformation of the sum of years since incorporation. Significant p-values are bold.

Table 5: Univariate Analysis of Differences variables between Firms that are below/equal and above the median value of BODFRANCO (2001-2016)

	BODFRA	BODFRANCO = <median (n="279)</th"><th>ANCO</th><th></th><th>Mann-</th></median>		ANCO		Mann-
	= <mediar< th=""><th colspan="2">>median (n=262)</th><th colspan="2">Whitney</th></mediar<>			>median (n=262)		Whitney
	Mean	Median	Mean	Median	p-value	p-value
Panel A: Dependent Variable						
PRICE	57.319	33.500	66.795	43.150	0.113	0.001
LnPRICE	3.312	3.497	3.312	3.765	0.000	0.001
Panel B: Independent Variable	es					
EPS	4.622	3.270	5.083	3.030	0.559	0.593
LnEPS	0.909	1.296	1.250	1.480	0.017	0.037
BV	6.691	10.000	7.683	10.000	0.003	0.000
LnBV	1.401	2.303	1.722	2.303	0.003	0.000
POLCON	0.484	0.000	0.240	0.000	(0.000)	
BODINDO	0.287	0.250	0.054	0.000	0.000	0.000
Panel C: Control variables						
LNSIZE	21.511	21.834	20.513	21.423	0.000	0.002
LEV	-1.055	-0.568	-1.126	-0.768	0.463	0.000
MANOWN	0.018	0.000	0.039	0.000	0.466	0.000
INSTOWN	0.518	0.596	0.513	0.573	0.793	0.379
LNAGE	3.753	3.689	3.966	3.761	0.001	0.008

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. The logged variables *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold.

	BODINDO =< median		BODI	BODINDO		Mann-
	value (n	n-282)	>median	>median (n=259)		Whitney
	Mean	Median	Mean	Median	p-value	p-value
Panel A: Dependent Variable						
PRICE	69.887	43.650	53.221	29.000	0.007	0.002
LnPRICE	3.557	3.695	3.450	3.611	0.322	0.586
Panel B: Independent Variables						
EPS	5.472	3.830	4.164	2.620	0.123	0.011
LnEPS	1.065	1.391	1.070	1.353	0.974	0.967
BV	7.449	10.000	6.869	10.000	0.106	0.008
LnBV	1.569	2.302	1.545	2.303	0.827	0.903
POLCON	0.262	0.000	0.479	0.000	(0.000)	
BODFRANCO	0.791	0.833	0.408	0.455	0.000	0.000
Panel C: Control variables						
LNSIZE	20.690	21.479	21.395	21.575	0.004	0.067
LEV	-1.124	-0.750	-1.052	-0.580	0.494	0.001
MANOWN	0.042	0.001	0.013	0.000	0.263	0.000
INSTOWN	0.520	0.584	0.511	0.596	0.644	0.873
LNAGE	3 938	3 738	3 767	3 714	0.011	0 180

Table 6: Univariate Analysis of Differences variables between Firms that are below/equal and

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. LnPRICE, LnEPS, LnBV, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold.

CNO.

Table 7: Main	Regressions	(2001-2016,	n=541)
---------------	-------------	-------------	--------

Variable	Expected						
	Direction	1		2		3	
INTERCEPT	?	1.641		-0.586		-0.556	
		<i>18.492</i>	***	-1.294		-1.308	
LnEPS	+	0.475		0.409		0.394	
		<i>13.897</i>	***	11.316	***	9.640	***
LnBV	+	0.228		0.245		0.351	
		8.620	***	8.89 7	***	5.862	***
POLCON	?			0.130		0.435	
				1.976	*	2.719	***
BODFRANCO	+			-0.309		-0.320	
				-4.266	***	-4.768	***
BODINDO	G			0.108		0.029	
				0.513		0.142	
LNSIZE	+			0.018		0.003	
				1.336		0.242	
LEV	- ()			0.035		0.029	
				1.613		1.478	
MANOWN	+			0.118		0.117	
				4.073	***	4.143	***
INSTOWN	+			0.293		0.356	
				2.628	***	3.225	***
LNAGE	+			0.413		0.463	
				7.957	***	8.919	***
CG2004	+			-0.122		-0.122	
				-1.625		-1.773	*
POLCON*EPS	-					0.022	
						0.403	
POLCON*BV	-					-0.180	
						-2.688	**
Industries Fixed	?	Yes		Yes		Yes	
Adj R ²		0.727		0.757		0.761	
F-statistic		142.393	***	83.100	***	76.3740	***

This table presents the result after adjusted heteroscedasticity and t-statistics(in paratheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold. . ***,**,* denote significance at 1%, 5% and 10% level, respectively (one-tailed).

Lubic of Sen Sciection 1 cs

r

Dependent variable	POLCON	PRICE		PRICE	
Variable	1	2		3	
INTERCEPT	0.305	-1.141		-2.055	
	1.072	-1.143		-1.712	*
BIG4	0.043				
	0.823				
LnEPS	0.028	0.434		0.459	
	2.022 **	7.385	***	7.150	***
LnBV	-0.024	0.227		0.310	
	-1.658 *	7.479	***	7.021	***
POLCON		0.131		0.477	
		1.983	**	2.634	**
BODFRANCO	0.260	-0.115		0.202	
	2.732 ***	-0.370		0.519	
BODINDO	0.957	0.917		2.201	
	8.802 ***	0.627		1.263	
LNSIZE	0.002	0.021		0.009	
	0.228	1.568		0.687	
LEV	-0.083	-0.037		-0.163	
	-5.770 ***	-0.279		-1.033	
MANOWN	-0.062	0.065		-0.027	
	-6.396 ***	0.671		-0.233	
INSTOWN	0.386	0.630		1.272	
	4.086 ***	0.972		1.604	
LNAGE	-0.082	0.337		0.263	
	-1.617	2.139	**	1.613	
CG2004	-0.035	-0.148		-0.196	
	-0.470	-1.979	**	-2.567	**
IMILLS		0.858		2.316	
		0.548		1.234	
POLCON*LnEPS				0.018	
				0.328	
POLCON*LnBV				-0.197	
				-2.582	**
Industry fixed	Yes	Yes		Yes	
McFadden R ² / Adjusted R ²	0.338	0.757		0.761	
LR/F statistic	14.424 ***	78.606	***	72.913	***
Obs with Dep=0	343				
Obs with Dep=1	198				

This table presents the result after adjusted heteroscedasticity and t-statistics (in paratheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity1. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is

Asian Review of Accounting

<text> percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the

Table 9: Regressions results on Political connections and Ethnicity (2001-2016, n=541)

2 Variable	Expected								
4	Direction	1		2		3		4	
5 Intercept	?	-0.586		-0.663		-0.703		-0.184	
6 ⁻ 7		-1.294		-1.604		-1.686	*	-0.441	
, 8 LnEPS	+	0.409		0.752		0.605		0.310	
9		11.316	***	5.861	***	6.002	***	4.879	***
10_{LnBV}	+	0.245		-0.020		0.134		0.253	
11		8.897	***	-0.146		1.214		3.282	***
13 POLCON	?	0.130		0.133		0.266		0.128	
14		1.976	*	1.971	**	1.251		0.617	
15 BODFRANCO	+	-0.309		-0.209		-0.319		-0.311	
16		-4.266	***	-0.932	**	-1.406		-2.414	**
18 BODINDO	-	0.108		-0.206		-0.011		-1.597	
19		0.513		-0.597		-2.214	**	-2.309	***
20 BODFRANCO*LnEPS	+			-0.358		-0.310			
21				-2.092	**	-2.114	**		
22 33BODFRANCO*LnBV	+			0.189		0.252			
24				1.129		1.661	*		
25 BODINDO*LnEPS	_			-0.586				0.761	
26				-3.741	***			2.082	**
27 28 BODINDO*LnBV	-			0.511				0.265	
29				2.449	**			0.653	
30 POLCON*LnEPS	?					-0.355		0.312	
31						-2.975	***	3.644	***
32 22 POLCON*LnBV	?					0.273		-0.269	
34						2.065	**	-2.820	***
35 POLCON *BODFRANCO	?					0.299			
36						0.844			
37 20 POLCON *BODINDO	?							1.564	
39								2.123	**
40 POLCON									
41 *BODFRANCO*LnEPS	?					0.639			
42						3.434	***		
⁴³ POLCON *BODFRANCO*LnBV 44	?					-0.746			
45						-4.091	***		
46POLCON *BODINDO*LnEPS	?							-1.251	
47								-3.320	***
⁴⁸ POLCON *BODINDO*LnBV 49	?							0.248	
50								0.585	
51 Control Variables		Yes		Yes		Yes		Yes	
52 Industry Fixed		Yes		Yes		Yes		Yes	
$53 A dj R^2$		0.757		0.768		0.778		0.785	
F-statistic		83.100	***	72.412	***	67.280	***	70.363	***

56

1

57 this table presents the result after adjusted heteroscedasticity and t-statistics(in parantheses). PRICE is share price of firm i for the period t in Mauritian 58 upees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. BODFRANCO is the percentage of Franco-50 Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. 53 Significant p-values are bold. ***,**,* denote significance at 1%, 5% and 10% level, respectively (one-tailed).

Table 10 Regression results for interaction between board ethnicity

	Bodfranco<	=med	Bodfranco>r	ned	Bodindo<=m	ned	Bodindo>m	ed
Variable	1		2		3		4	
INTERCEPT	-0.188		3.280		-1.847		0.023	
	-0.301		3.272	***	-1.480		0.037	
LnEPS	0.407		0.203		1.421		0.298	
	3.825	***	3.154	***	4.244	***	2.959	***
LnBV	0.127		0.257		-0.426		0.323	
	0.901		2.989	***	-1.737	*	3.002	***
POLCON	-0.407		0.622		0.248		0.367	
	-1.452		2.271	**	0.235		1.244	
BODFRANCO	-0.778		-3.002		0.014		-1.416	
	-3.074	***	-4.828	***	0.053		-3.807	***
30DIND0	-0.583		-8.872		0.492		0.370	
	-0.610		-3.975	***	0.380		1.338	
BODINDO *LnEPS	-0.138		2.197					
	-0.318		2.252	**				
BODINDO *LnBV	0.704		1.759					
	1.227		1.276					
BODFRANCO *LnEPS					-1.244		0.058	
					-3.183	***	0.338	
BODFRANCO *LnBV					0.808		0.553	
					3.111	***	2.559	**
POLCON*LnEPS	0.284		0.188		-0.480		-0.014	
	2.131	**	1.810	*	-1.043		-0.108	
POLCON*LnBV	0.009		-0.291		0.016		-0.101	
	0.051		-2.606	***	0.027		-0.621	
POLCON*BODINDO	1.443		-2.564					
	1.383		-0.418					
POLCON*BODFRANCO					0.451		-0.360	
					0.366		-0.341	
POLCON*BODINDO *LnEPS	-0.425		-1.438					
	-0.939		-1.210					
POLCON*BODINDO *LnBV	-0.430		0.711					
	-0.725		0.211					
POLCON*BODFRANCO *LnEPS					0.696		0.231	
					1.306		0.968	
POLCON*BODFRANCO *LnBV					-0.402		-0.119	
					-0.643		-0.229	
Control variables	Yes		Yes		Yes		Yes	
Industry fixed	Yes		Yes		Yes		Yes	
Ν	279		262		282		259	
Adj R ²	0.865		0.732		0.767		0.836	
<i>F-statistic</i>	65.456	***	26.187	***	30.764	***	55.266	***

Asian Review of Accounting

<text><text> This table presents the result after adjusted heteroscedasticity and t-statistics(in parantheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors.

Significant p-values are bold. ***, **, * denote significance at 1%, 5% and 10% level, respectively (one-tailed).

The Mauritius Commercial Bank L	td
The Mauritius Development Invest	ment Trust Co Ltd
Omnicane Ltd	
United Basalts Ltd	
Mauritius Stationery Manufacturers	s Ltd
Mauritius Chemical and Fertilizer I	ndustry Ltd
Rogers & Co. Ltd	
Swan Insurance Co. Ltd	
Ferra Mauricia Ltd	
Compagnies des Magasins Populai	res Ltée
Vivo Energy Mauritius Ltd	
Sun Resorts Ltd	
Mauritius Union Assurance Co. Lto	I L
Mauritian Eagle Insurance Co. Ltd	
Automatic Systems Ltd	
Air Mauritius Ltd	
State Bank of Mauritius Ltd	
nnodis Ltd	
Go Life International Ltd	
Alteo Ltd	
Cim Financial Services	
Bramer Banking Corporation	

Responses to Editor Comments, Suggestions and Recommendations on ARA-10-2022-0238.R1

Dear Prof. Nan Zhou,

Thank you for considering our paper for publication in Asian Review of Accounting. We are grateful to the editor and reviewer for their time and constructive comments on our manuscript. These have greatly helped us to improve the quality of our manuscript. We have implemented their comments and suggestions and we are grateful that the reviewer is satisfied with our revision.

The editor's comments are reproduced and our responses follow each of the comments and suggestions.

1. Your manuscript has not cited any articles from Asian Review of Accounting (ARA). To increase the impact of the journal, please cite three articles from ARA.

Authors' response:

The following papers were already included in the manuscript:

- Abdul Wahab, E.A., Pitchay, A.A. and Ali, R. (2015), "*Culture, corporate governance and analysts forecast in Malaysia*", Asian Review of Accounting, Vol 23 No. 3, pp. 232-255. <u>https://doi.org/10.1108/ARA-03-2014-0033</u>
- Boonlert-U-Thai, K. and Schaberl, P. (2022), "Value relevance of book values, earnings, and future earnings: Evidence by time, life cycle stage, and market uncertainty". Asian Review of Accounting, ahead-of-print.

Therefore, the following source has been added as a third paper from ARA:

Tantawy, S.M. and Moussa, T. (2023), "The effect of political connections on firms' auditor choice decisions and audit opinions: evidence from Egypt", Asian Review of Accounting, ahead-of-print. https://doi-org.ezproxy.mdx.ac.uk/10.1108/ARA-07-2022-0161

Please refer to the first paragraph of the revised manuscript.

2. Please cross check the references to make sure papers cited in the text are in the reference section and vice versa.

Authors' response:

This has been completed. References have both an in-text citation and an out—text citation. The references have also been updated using the ARA referencing style. References without an in-text citation have been removed from the reference list. Each DOI and website links have

been checked and updated where needed. Please refer to the references section in the revised manuscript.

3. Please read a few recent ARA articles and carefully prepare your paper following the ARA format. This applies to tables, references, section titles, etc.

Authors' response:

This has also been completed. The manuscript has been formatted using the most recent papers published in ARA. The Abstract has been amended to follow the guidelines of ARA.

4. Please have your paper copy-edited, if you haven't done so. If you have, please indicate in the response letter.

Authors' response:

We have sent our paper for professional copy editing. Further, we have made several changes to the paper for ease of reading. Please refer to the track-changed version of the manuscript.

r Rounting

We look forward to the outcome of your assessment.

Yours sincerely,

Effiezal Aswadi Abdul Wahab

(Corresponding author)

Political connections, board ethnicity and value relevance in Mauritius

<u>Abstract</u>

Purpose - The paper aims to investigate the impact of political connections and board diversityethnicity on the value relevance of earnings and book value in Mauritius.

Design/methodology/approach - This study is based on a sample of 541 Mauritian listed firmyear observations for 2001-2016. Financial and board diversity data have been collected using the listed firms' annual reports and from reports published by the Stock Exchange of Mauritius. Political connection data was derived from the directory of Chief of State and Cabinet members. The research hypotheses were empirically tested with the use of using a modified Ohlson (1995) price model.

Findings - This study shows that political connections negatively impact the value relevance of earnings and book value. We find that firms that have Franco-Mauritian directors will constraint the negative impact of political connections with Franco-Mauritian directors will constrain political connections' negative impact. while Wwe find contrasting results for Indo-Mauritian directors since they form an integral part of the government in Mauritius.

Originality - This study contributes to the scarce accounting literature in the context of Mauritius. Firstly, to date, no study has investigated the relationship between the value relevance of accounting information and political connections in Mauritius. Secondly, Mauritius' capital market is dominated by a non-indigenous ethnic group, Franco-Mauritians, who remains the economic elite. Hence, Mauritius presents an opportunity to bring forth another important aspect in the capital market and corporate governance; diversity on the board of directors. Therefore, the study extends onto the political connections and board diversity literature.

This paper investigates the impact of political connections and board diversity on the value relevance of earnings and book value in Mauritius. Based on a sample of 541 Mauritian listed firm-year observations for 2001-2016, this study shows that political connections negatively impact the value relevance of earnings and book value. We find that firms with Franco-Mauritian directors will constrain the negative impact of political connections. We find contrasting results for Indo-Mauritian directors since they form an integral part of the government in Mauritius. Overall, this study contributes to the extant literature on political connections coupled with the context of ethnicity in a developing capital market. The research hypotheses were empirically tested using a modified Ohlson (1995) price model, and the results are robust after controlling for selection bias.

Keywords_; political connections, board diversity, ethnicity, value relevance, Mauritius.

JEL classifications: G3, G14, M14, M41

Paper type Research paper



JEL elassifications: G3, G14, M14, M41

1. Introduction

Several studies have examined the economic consequences of political connections in capital markets. Studies on political connections center, but are not limited to, the quality of accounting information (Chaney *et al.*, 2011), audit fees (Gul, 2006), capital and resources allocations (Johnson and& Mitton, 2001), financial analysts (Chen *et al.*, 2010; Gist and& Abdul Wahab, 2021), firm performance (Fisman, 2001), auditor choice (Guedhami *et al.*, 2014; Tantawy and Moussa, 2023), cost of equity capital (Boubakri *et al.*, 2012) and corporate lending (Khawaja and& Mian, 2005).

The findings of these studies are rather mixed, suggesting that political connections could be detrimental or not to capital markets. On one hand, these studies suggest that political connections or politically connected firms are rent-seekers and promote cronyism and nepotism. On the other hand, conventional wisdom tells us that being connected would provide capital funding via grants and increase performance (Fisman, 2001). In addition, the stream or resources from the income for connected firms are fairly-relatively uncertain, volatile, and require a higher level of higher scrutiny.

Mauritius's political connections <u>andor</u> earnings quality studies are limited and in the infancy stage. The extant literature in Mauritius has primarily focused on the implementation of the code of corporate governance (Mahadeo <u>and</u>& Soobaroyen, 2016<u>3</u>), the investigation of board diversity (gender, age, and education), and corporate governance (Munisi & and Randoy, 2013), firm performance (Mahadeo *et al.*, 2012), and a case study investigating the role of shareholders activism in Mauritius (Beebeejaun and& Koobloll, 2018).

We extend these studies on several grounds. First, none of the above studies has examined political connections, an <u>important essential</u> feature in a developing capital

market like Mauritius. Given the interest, we investigate whether political connections affect the value relevance of accounting information in Mauritius, as no study, to date, has investigated this relationship. Mauritius is an emerging economy in the African region, and the findings provide insight into the nature of connections in the country. Given the unique background that shapes its capital market, Mauritius provides an interesting avenue. We are driven by several high-profile ongoing cases surrounding politicians and heads of government. For example, soon after the December 2014 elections, the previous Head of Government, Dr. Navinchandra Ramgoolam, was arrested based on money laundering charges when two safes with billions of rupees and dollars were discovered at his residence (Wan, 2015).¹ In March 2016, the Prime Minister, Sir Anerood Jugnauth, asked the Minister of Environment to step down after bribery allegations were made against him (Reuters, 2016). These cases indicate that political connections are prevalent and influential in Mauritius, questioning the linkage between political board members and the value of published accounting numbers.

Secondly, the Mauritius capital market is dominated by a non-indigenous ethnic group, Franco-Mauritians, who remain the economic elite (Carroll and Carroll, 2000). Hence, Mauritius presents an opportunity to bring forth another important aspect of capital markets and corporate governance; diversity on the board of directors. Board diversity is important in promoting good governance (Carter *et al.*, 2010). The guiding ethics behind board diversity is to promote governance and better monitoring by providing helpful insight, obtaining external resources, expertise, and exercising diligence (Simkins and Carter, 2003; Srinidhi *et al.*, 2011; Tee, 2019). Consequently, in theory, these traits enhance firm value and the value relevance of accounting information (Ntim, 2013). However, this is not always the casrue when coupled with a

nation's socio-economy disparity (Dargar and Joomun, 2005; Mishra and Jhunjhunwala, 2013).

A distinctive feature of the Mauritius capital market is that it is dominated by white Franco-Mauritians (Mauritians of White-French descendants), who only constitute around two percent of the total population. The other ethnic group, Indo-Mauritians, does not dominate the capital market despite forming almost 68 percent of the population (Srebrnik, 2002; Bunwaree and & Kasenally, 2005). This 'disparity' or 'mismatch' between the ethnic groups and Franco-Mauritians' dominance in the capital market presents interesting avenues for research in this area. This landscape is not unique to Mauritius, as other countries, such as Malaysia, experience a similar situation. However, the dominance of the non-majority Chinese-Malaysian is not as prevalent as Franco-Mauritians in Mauritius. Mauritius has not had a question regarding ethnicity in the national census since 1972; however, Indo-Mauritians consist of two-thirds of the total population (CIA World Factbook, 2021), and several researchers refer to the following statistics: the Indo-Mauritians 68% (Hindus 52% and Muslims 16%); Creoles Mauritians 28%; Sino-Mauritians 3% and Franco-Mauritian less than 2% of the population (Srebrnik, 2002; Bunwaree and& Kasenally, 2005; Salverda, 2015). We opted for these two board ethnicity measures as they represent 'extremes' (Salverda, 2015) as one group (Franco-Mauritians) is dominant in the capital market (in this case, board participation). At the same time, Indo-Mauritians are a minority in the capital market, although they have the largest population. These differences present us with competing hypotheses to test.

We predict that white Franco-Mauritian will mitigate the negative impact of political connections on the value relevance of book value and earnings. The premise of our argument is simple. The white Franco-Mauritians have has been a dominant player in the capital market and provide, providing diverse expertise, know-how, and, more

importantly, experience. Their presence on the board of directors provides these characteristics that could offer a monitoring role and increase earnings quality. The white Franco-Mauritian are not known to be active participants in the political scene in Mauritius. One would say that their dominance in the capital market presents a form of check and balance to Mauritius's politicians or political connections. Since white Franco-Mauritians are not active in politics, their survival in the capital market could signal their ability to tap into the external capital market and demonstrate the ability to sustain themselves without government assistance (Salverda, 2015).²

In contrast, we predict that political connections will negatively impact the relevance of book value and earnings for Indo-Mauritians. The existing literature supports this view. Although dominant in the political scene, Indo-Mauritians are less prominent in the capital market. They often use their presence in the capital market as a symbol, mainly for attracting funding or government grants (Salverda, 2015). For example, the presence of Malay directors in Malaysia during the early stages of independence was largely ceremonial and served as an indicator of exposing government servants to the capital market (Abdul Wahab *et al.*, 2018). In contrast, we predict that the negative impact of political connections on the value relevance of book value and earnings will worsen for Indo-Mauritians. Indo-Mauritians are dominant in the political scene but not in the capital market. Their presence in the capital market could be symbolized as rather ceremonial (Salverda, 2015) to attract funding or government grants. The extant literature supports this view. For instance, the presence of Malay directors in the early stages of independence has been seen as largely ceremonial and an indicator of exposing government servants to the capital market (Abdul Wahab *et al.*, 2018).

Based on a sample of 541 firm-year observations of Mauritian listed firms (SEM) for 2001-2016, we find that both the book value and earnings are value relevant in the capital market. As predicted, we find the value relevance of book value is lower for politically connected firms. This finding is consistent with the argument that politically connected firms have lower earnings quality and create uncertainty in the market. Furthermore, the negative impact of political connections is compounded by the recent high-profile cases involving local politicians, as mentioned above.

The results confirm our predictions: the presence of Franco-Mauritian directors mitigates the negative impact of political connections on the value relevance of earnings. This suggests that Franco-Mauritian directors mitigate agency issues and monitor and perform checks and balances on political connections. In support of our arguments, we find that Indo-Mauritians' directors increase the negative impact of political connections on value relevance. The results are not surprising since Indo-Mauritians dominate the government as politicians and governmentunsurprising since Indo-Mauritians dominate the government as politicians and officers. After controlling the selection test and several additional analyses, the results remain qualitatively similar.

The findings of this study contribute to the literature on political connections, ethnicity, and value relevance in the following ways. First, the paper extends the evergrowing literature on political connections, especially in Africa. Given Mauritius's unique political, legal, and social backgrounds relative to other African countries, the findings of this study act as a catalyst for <u>other-further</u> political connection studies.

The results of this study present another view on how ethnic groups interplay with the premise of political connections. The institutional background of Mauritius allows us to examine the different dimensions of political connections in the capital market. Second, the study tackles ethnicity and provides an exciting outcome on how ethnic groups play

 their role in developing the capital market. Third, the findings of this study contribute to the understanding of how the various ethnic groups in a capital market play a role.

The paper is organized as follows. Section 2 summarizes the political and social background in Mauritius. Section 3 provides the rationale behind the hypotheses developed. Section 4 presents the methodology and sample selection. The results and robustness tests are discussed in Section 5, while section 6 discusses the additional analysis. Section 7 concludes the paper.

2. Institutional background

2.1. Political Economy in Mauritius

Since 1810, British governors and officials administered the island of Mauritius. In the 1960s, the negotiations for independence started, and elections were held in 1967. Consequently, Mauritius became a fully-fledged democracy and a Commonwealth member on 12 March 1968. The independent state was then run by the Labour Party and two smaller parties: The Independent Forward Bloc and the Muslim Action Committee – two ethnic-based parties. Since 1979, the elections have resulted in coalition governments with the same political parties rotating from the government to the opposition. Therefore, community-based favoritism and nepotism remain a source of public frustration even though corruption is not customary as per regional standards.³

The political structure in Mauritius was adapted from the British Empire and implemented after independence in 1968.⁴ In line with its globalization strategy, since the 1980s, the government of Mauritius has taken the role of a business facilitator to create a positive business environment on the political, economic, and social levels. The aim is to nourish and expand the private sector. Furthermore, the government has controlling shares in numerous listed firms: Air Mauritius – the national airline; State Bank of

Mauritius; and Mauritius Telecom – the telecommunication company. The government also invests in a broad range ofvarious corporations via its investment entity – the State Investment Corporation. The board of directors' positions are primarily allocated to senior government officials, with the chairperson nominated by the ruling government. The chairpersons are rotated once there is a change in government. The new ruling party has the authority to remove a chairperson nominated by the previous government and replace him/her with a new chairperson fitting the ideologies of the new ruling party.

However, Franco-Mauritians have retained major land ownership; over time, they have created family holdings to manage those lands (Salverda, 2013). The family holdings still exist and are majorly quoted on the SEM. The boards of directors of those listed firms are composed of directors drawn from the same small and tightly-knit white Franco-Mauritian business community. The directors are dominant male directors who have been appointed to numerous boards for decades (Mahadeo *et al.*, 2012).

Mauritius has no indigenous population but is known for its cultural diversity. As Mahadeo and Soobaroyen (2013) mentioned, Mauritius has a dual colonial past being both an ex-French and British colony, and therefore its population came mainly from European colonialism and immigration. Also, Brautigam (2009) argues that even though the population originates from the four corners of the world – Europe, Africa, India, and China, world's four corners – Europe, Africa, India, and China- it has retained close links with French and British traditions.

In 1835, Mauritius' geographical setting was dominated by sugar cane plantations showing early signs of a one-crop economy that relied on slave labour (Holmberg, 1962). However, after the abolition of slavery in the British Empire in 1834, the demand for sugar cane plantations' labour increased, especially with preferential prices being introduced by Britain on Mauritian sugar. The <u>demand-need</u> for such labour led to the

"Great Experiment" initiated by the British government. More than 450,000 labourers were expatriated from India to Mauritius, with the status of indentured labourers working in Mauritius' sugar cane fields. Most of today's Indo-Mauritians, of Hindu or Muslim faiths, are descendants of these indentured labourers, while a minority came freely as traders or educators (Chiriyankandath, 2009). Two-thirds of these Indo-Mauritians remained permanently on the island, while one-third returned to their home country or migrated to other British colonies (Aapravasi Ghat Trust Fund, 2015).

2.2. Social background and its impact on financial reporting

Indeed, while the Mauritian population is diverse, property and business ownership of the largest firms remain concentrated and is traceablthe largest firms' property and business ownership remain concentrated and traced back to the heritage left by colonization (Lange, 2003). The World Bank (20120) reports that white Franco-Mauritian families control between 5 and 7 large listed firms. These families, which benefited mainly from the sugarcane industry, consequently invested massively in other local industries. This can also be observed in the directors' composition: most of the directors are white Franco-Mauritian directors, while some minority directors come from the Creole community, the Indo-Mauritian community, and the Sino-Mauritian population and the diversity of the boards of directors. While many directors appear to have been appointed for their knowledge, skills, and expertise, some directors are appointed due to connections to family and majority shareholders. Therefore, those family-dominated management structures have a definite impact on the boards of SEM-listed firms.

Furthermore, the majority shareholder usually appoints non-executive directors, which questions the independence of those nominated. This again creates an interesting

setting to probe the influence of board diversity on published accounting information's value relevance.

3. Hypotheses development

3.1. Value relevance in the Stock Exchange of Mauritius (SEM)

In the accounting literature, value relevance has been defined<u>The accounting literature</u> defines value relevance as the association between company values and accounting numbers (Ohlson, 1995; Francis and Schipper, 1999; Barth *et al.*, 2001; <u>Kothari, 2001;</u> Beisland *et al.*, 2010). Value relevance research investigates the usefulness of reported numbers to accounting information users; and current and potential investors.

Ball and Brown (1968), Beaver (1968), and Beaver *et al.* (1980) explore a plausible association between accounting variables and share prices. Beaver (2002) explains that value relevance research examines how accounting variables relate to a dependent variable based on security price. Negakis (2005) believes that this research aims to assess how helpful accounting information is for users, like investors.

Beaver (2002) states that value relevance research tests the associations between a set of accounting variables and a security price-based dependent variable, while Negakis (2005) argues that the primary goal of existing value relevance research is to investigate the usefulness of reported accounting numbers to users of accounting information such as investors. Indeed, investors widely use financial reports to assess the market value of firms for investment decisions (Adedeji and& Kajola, 1999). Consequently, it can be seen that the main objective of financial reporting is to provide equity valuation information to potential investors. This objective is empirically analysed analyzed by value relevance research. This justifies the study of the_value relevance of accounting information in Mauritius; henceforth subsequently leading to the first hypothesis:

*H*₁: Accounting information produced by SEM-listed firms is value relevant.

Badu and Appiah (2018) studied the value relevance of accounting information of public firms using Ohlson's price model (1995) applied on the Ghana Stock Exchange for the period-2005-2014 using a sample of 224 firm-year observations. They find a significant and positive relationship between earnings and book values of equity to stock prices. Furthermore, the same conclusion is reached by Diftar and Elkalla (2019), who examined the value relevance across the different countries of North Africa and the Middle East from 2007 to 2016.

3.2. Political connections and value relevance

According to Faccio (2006), a firm is politically connected if one of the firm's largest shareholders or senior executives is a member of parliament (MP), a minister, or the head of government, or closely related to a top official. A large shareholder is defined as anyone who controls a minimum of 10% of the shareholders' votes directly or indirectly (Faccio 2006). Faccio (2006) also includes the political connections through a relative, whereby one relative of a head of state or minister is a large shareholder or a management executive. Firms are also classified as politically connected if: (i) A head of state or one of his/her relatives was also a top executive or MP during the study period; (ii) A current government minister or MP was a senior executive or significant shareholder of a company in the past; (iii) A top executive or substantial shareholder is a politician of another country; (iv) A substantial shareholder or senior executive is known to be associated with a political party whether the ruling parties or opposition parties (Civilize *et al.*, 2015). Other researchers classified state ownership as a form of political ties with firms and special shares held by the government (Jones *et al.*, 1999; Bushman *et al.*, 2004; Nee *et al.*, 2007; Hanousek *et al.*, 2007).

There are many ways for firms to obtain benefits from political connections. Such connections can lower tax burdens (Adhikari *et al.*, 2006), increase the preferential treatment to financing (Dinc, 2005) and political bailouts in the event of financial distress (Faccio *et al.*, 2006), and result in a higher allocation of government investment during a financial crisis (Johnson and Mitton 2003). Chaney *et al.* (2011) suggest that political connections should increase earnings quality due to heightening media scrutiny, leading to increased firm monitoring. With increased scrutiny, better access to resources, and enhanced monitoring due to the public or state interest, these connected firms should have higher levels of earnings quality relative to non-connected firms.

Batta *et al.* (2014) investigate whether a country's degree of expropriation risk could amount to a positive rapport between political connections and earnings quality. Batta *et al.* (2014) argue that countries with high expropriation risks will lower their quality of earnings, but not for politically connected firms since the risk is lower due to connections. Even though Mauritius has a low ranking in terms of expropriation risk, in 2014, the government took over the BAI Group, a leading conglomerate in Mauritius. The group's assets were seized and sold to local firms, while some were retained under the government's control. Moreover, Fung *et al.* (2015) similarly claim that the duration of a connection could suggest that the connected firm may self-sustain into the future and thus could increase earnings. Political connections create some form of certainty, especially in dealing with resources. The above suggests a positive affiliation between political connection and earnings quality; hence a positive affiliation between political connection and the value relevance of accounting information is a component of earnings quality. Conventional wisdom suggests that political connections or relationship-based economies provide an advantage to the connected firms to capital (Fisman, 2001) and

information from the government on policy changes. Such advantages would improve the market value of connected firms relative to unconnected firms.

A classic example was from the work of Johnson and Mitton (2003) when they examined the impact of capital control on connected firms in Malaysia during the Asian Financial Crisis of 1998-1999. They uncovered that capital control restricted the outflow of funds and assisted or propped the connected firms during the crisis. Although the capital control period indicates that the connected firms were operating inefficiently before the crisis, such policy (e.g., restriction of the flow of funds) increases the value of connected firms.

Alternatively, existing literature suggests that political connections could be detrimental to the firmharm the firm's value and earnings quality. Political connections create agency costs between agent and principal and minority and majority shareholders (Faccio, 2006). The agency costs are owed to the high rent-seeking activities by the directors with political connections and the low level of transparency (Ball *et al.*, 2003). The high agency cost increases information asymmetry and creates uncertainty, resulting in decreaseddecreasing earnings quality. Chaney *et al.* (2011) offer three reasons to contribute to a negative relationship between political connections and the quality of earnings. First, insiders of connected firms could hide, obscure, or delay reporting the advantages received to deceive investors. Second, Chaney *et al.* (2011) argue that the connected firms do not worry about the quality of accounting information as politicians shield them; and the third argument is that firms with poor earnings quality are more inclined to establish political connections. Therefore, the political connection appears to have an ambiguous impact on investors, and thus, this leads to the following research hypothesis:

H₂: Political connections weakens the value relevance of book value and earnings
The evidence examining the relationship between political connections and earnings quality is mixed. Indeed, Chen et al. (2010) investigate the relationship between analyst forecast accuracy and political connections and find a negative relationship. They argue that uncertainty in the revenue-generation process of connected firms is highly questionable. Gul (2006) investigates the relationship between audit fees and political influences and finds a positive relationship. Gul argues that auditors perceived connected firms as risky due to the high agency costs. Faccio (2006) shows no significant price effect when a politician is appointed to corporate boards, yet the share price increases when an executive enters politics. On the same line, many authors establish that earnings quality is systematically lower for firms with political connections (Chaney et al., 2011; Riahi-Belkaoui, 2004), and Leuz and Oberholzer-Gee (2006) find a reduction of foreign financing in politically-connected firms. Fisman (2001) shows that investors reacted to every rumour on-about the health of President Suharto. Thus, the share returns of politically connected Indonesian firms are noticeably lower than those of unconnected firms. The same is confirmed by Bertrand et al. (2004); Fan et al. (2007); Li et al. (2008); and Civilize et al. (2015) in the French, Chinese, and Thai contexts, respectively.

3.3. Political connections, board ethnicity, and value relevance

The agency theory suggests that board diversity increases the independence of the board as the differences in gender, ethnicity, and background trigger additional questioning of executive's decisions (Arfken *et al.*, 2004; Johnston and& Malina, 2008; Carter *et al.*, 2010; Lincoln and& Adedoyin, 2012; Abdullah, 2013; Triana *et al.*, 2014). As mentioned by the human capital theorythe human capital theory mentions, board diversity brings increased creativity and innovation, new ideas and perspectives, and new skills and knowledge, which will greatly help decision-making (DiStefano and& Maznevski, 2000;

Baranchuk <u>and</u> Dybvig, 2009; Luckerath-Rovers, 2013). <u>Diversity on boardsBoard</u> <u>diversity</u> can bring the company closer to its stakeholders and create a more general responsibility towards society (Van Der Walt <u>and</u> Ingley, 2003).

Consequently, several interdisciplinary theories strongly indicate a link between a firm financial performance and board diversity. Human capital and resource dependence theories suggest a positive relationship between firm performance and board diversity (Carter *et al.*, 2010). Yet, the agency theory advises that the nature of the relationship might not be explicit even though proper management monitoring may drive higher firm value (Ntim, 2013).

The appointments of directors based on ethnic grounds could be for several reasons. Resource dependency theory suggests that such appointments are based on human and social capital needs. These appointments are usually based on special skills only obtained by certain ethnic groups (human capital) or the needs of firms to seek economies of scale via networking (social capital). Many studies state that the board of directors remains one of the utmost crucial subsets within an organization (Lipton and&e Lorsch, 1992; Sonnenfeld, 2002; Bart and&e McQueen, 2013) as it performs many strategic functions within the organization (Bilimoria and&e Piderit, 1994; Lincoln and&e Adedoyin, 2012; Dale-Olsen *et al.*, 2013; Ntim, 2013). In addition, Agarwal and Knoeber (2001) argue that firms acquire different advantages and resources when they appoint ethnic minority directors, and boards which haveith several dealings with the government or face government intervention and regulation are more likely to appoint the same directors.

Overall, $t\underline{T}$ he studies demonstrate that ethnic diversity enhances the corporate governance element of board independence, monitoring executives' decision-making, and acquiring resources, consequently improving the firm's value (Ntim, 2013). Furthermore,

ethnic directors provide unique information to the board, which provides better information to the executives and improves decision-making (Abdullah, 2013).

Oppositely, Goodstein *et al.* (1994) explored board diversity's impact on decisions involving strategic changes, and they concluded that homogenous boards tend to initiate strategic changes more often than diverse boards. This implies that diverse boards might negatively impact firm valuation. Several researchers explain that directors from ethnic minorities might be chosen by tokenism; to tick a box imposed by the government (Hillman *et al.*, 2007; Abdullah, 2013; Ntim, 2013). Finally, a diverse board may increase conflicts between board members as they bring their interests and loyalties to the board (Roberson and& Park, 2007; Wellage and& Locke, 2013), preventing board cohesion (Goodstein *et al.*, 1994; Lincoln & Adedovin, 2012).

Preceding studies have indicated that political affairs may impact board composition (Agarwal and& Knoeber, 2001; Fan *et al.*, 2007; Chen *et al.*, 2010), as directors with political connections bring several advantages. Agarwal and Knoeber (2001) suggest that outside directors who have political connections can assist the firm in navigating political affairs by using their abilities to anticipate or influence government actions. These abilities can come from their previous involvement in government or political parties and their experience with legal proceedings against the government. Agarwal and Knoeber (2001) argue that outside directors with political ties might help with the political dealings of the firm through their skills to predict or impact government actions, skills they have acquired from prior participation in government and political parties, or acquired through administration or legal proceedings against the government. Domadenik *et al.* (2016) mention that a government might use its power illegitimately to appoint people without adequate competencies to boards of state-owned firms as long as they belong to the political network. The appointed directors are

positioned as a special purpose vehicle to extract rent or are being rewarded for previous political engagements.

From an accounting point of view, diverse boards, especially ethnically diverse boards, are expected to lead to better performance and higher earnings quality through enhanced corporate governance and accountability (Srinidhi *et al.*, 2011; Tee, 2019). Furthermore, Kim *et al.* (2013) studied the effect of boards with diverse political ideologies on firm performance. They concluded that boards with political diversity are linked with higher firm performance (Liang *et al.*, 202149). While the literature is scarce in this field, it might be applieablen this field is scarce, it might apply in Mauritius. Indeed, political connection measures the relationship between firms and political parties. However, in Mauritius, adherence to political parties or simply maintaining a relationship with a specific party is closely related to the ethnic group to which the person belongs.

We argue that board ethnicity, our proxy for board diversity, would moderate the negative impact of political connections on the value relevance of earnings and book value. The premise of the moderating variable is simple, and one would expect the positive impact of board diversity would mitigate the negative relationship between political connections and <u>the</u> value relevance of earnings and book value.

However, the institutional background in Mauritius would provide a different story. <u>Our study focuses on two ethnic groups: Franco-Mauritians, primarily white and</u> <u>descended from French and British colonizers, and Indo-Mauritians. The two ethnic</u> groups in our study are Franco-Mauritians, who are dominantly white, and descendants of French and British colonizers, and Indo-Mauritians. They originated from India and reached Mauritius as indentured labourers, merchants, traders, and high-profile teachers and officers under British rule. The Franco-Mauritians group, representing a minority ethnic group in Mauritius, is dominant indominates the local capital market. However, despite having massive control in the capital market, they are hardly present as officers or local representatives in the government. We expect that the negative impact of political connections on the value relevance of earnings and book value is weakened by the presence of Franco-Mauritian directors. This scenario presents a 'check and balances' to the government's influence in the capital market. In addition, their presence in the capital market signals their ability to tap into the external market. Based on the above, we predict the following hypothesis:

 H_3 : The negative impact of political connections on value relevance will be weakened by Franco-Mauritians' directors

The appointment of Indo-Mauritian directors who dominate the political scene could be seen as ceremonial rather than as playing an active role in monitoring. In contrast, we expect that the presence of Indo-Mauritian directors will not mitigate the negative impact of political connections on the value relevance of earnings and book value. Carroll and Carroll (2000) highlight that the government deliberately expanded public employment in the 1970s by hiring unskilled and unemployed, mostly Indo-Mauritians. This has since escalated to ministries and parastatal organisations-organizations (Carroll and & Carroll, 2000). In anecdotal evidence supplied by Carroll and Carroll (2000), employment in these institutions is presented as a reward to supporters and clients. Hence, in line with prior theories and studies, the following is hypothesized:

 H_4 : The negative impact of political connections on value relevance will be heightened by Indo-Mauritian directors

Several studies have examined this relationship in a similar setting. Studies have documented evidence in a similar setting. Abdul Wahab *et al.* (2015) investigate the

relationship between culture and analyst forecast accuracy in Malaysia. According to them, Bumiputera directors, the largest ethnic group in Malaysia but not as prominent in the capital market, have lower governance levels, less disclosure capacity, and more inherent risk and inefficiency due to preferential treatment. This will lead to an increase in agency costs. They suggest that the Bumiputera directors, who are the dominant ethnic group in Malaysia, but less dominant in the capital market—and who have a lower level of governance, fewer disclosure capabilities, and carry a higher level of inherent risk and inefficiency due to preferential treatment—will increase the agency cost. Based on this premise, they argue that the increased agency cost will increase information asymmetry, impacting analysts' forecasts. Abdul Wahab *et al.* (2015) find results that support their arguments.

4. Methodology

4.1. Sample and Data

The initial population of the study is the firms listed on the SEM from 2001 to 2016. The number of firms listed during the sample period is 43, resulting in 688 firm-year observations. We exclude financial firms as they are subject to different accounting regulations and risk structures. We exclude the three (3) financial firms, which result in 48 firm-year observations. Due to missing financial data, we exclude 99 firm-year observations. Hence, the final sample for this study is 541 firm-year observations, as shown in Table 1.

(Table 1 here)

Financial data has been collected using the Thomson Reuters DataStream database, Bureau van Dijk's Osiris database, and mostly the listed firms' annual reports. The stock prices were gathered from reports published by the Stock Exchange of Mauritius. Political connection data has been derived based on the directory of Chief of State and Cabinet members of Foreign Countries, compiled by the Central Intelligence Agency of the United States of America. Finally, data used to identify board diversity was gathered primarily using the firms' annual reports and company websites.

The value relevance of accounting information is indicated by a statistical association between stock prices or stock returns and a set of accounting information (Beaver, 2002; Liu and Liu, 2007). Indeed, value relevance research intends to provide empirical data displaying the degree to which accounting numbers shape equity valuation and subsequently aims to assess the usefulness of these numbers to investors' decision-making.

Therefore, value relevance is usually tested using regression analysis: price regression or return (price change) regression. Early research in value relevance started with price regression, examining the relationship between the book value and the market value of equity. It is usually denoted as follows:

$$P^{it} = \beta_o + \beta_I B V^{it} + \varepsilon^{it}$$

Where:

 P^{it} = Market value or price per share of firm *i* at time *t*

 BV^{it} = Book value of total equity or Book Value per share of equity of firm *i* at time *t*

Furthermore, Ohlson (1995) shows that firm's value is a linear function of its book value, earnings and other relevant information (Daniel W-Collins *et al.*, 1997; Kim and Kross, 2005). Hence, the formal regression of this study is as per the following equation:

$$P^{it} = \beta_o + \beta_I EPS^{it} + \beta_2 BV^{it} + \varepsilon^i$$

 EPS^{it} = Earnings per share of firm *i* for the period *t*

Glasscock et al. (2021) shed light on the ongoing debate for proper scaling in empirical accounting research. They believe that using alternative specifications, other than scaling, may be more effective in addressing heteroscedasticity issues. In line, Lubberink and Willett (2021) and Boonlert-U-Thai and Schaberl (2022) argue that log-linear regression models are appropriately suited and provide valid estimates of the relationship between price and accounting data. Therefore, this study uses a log-linear model of Ohlson's (1995) model to estimate the value relevance of accounting information:

 $LnP^{it} = \beta_o + \beta_1 LnEPS^{it} + \beta_2 LnBV^{it} + \varepsilon^{it}$

Where:

 $LnP^{it} = Log \text{ of market value or price per share of firm } i \text{ at time } t$ $LnEPS^{it} = Log \text{ of Earnings per share of firm } i \text{ for the period } t$ $LnBV^{it} = Log \text{ of Book value of total equity or Book Value per share of equity of firm } i \text{ at time } t.$

In line with the log-linear regression models used in Boonlert-U-Thai and Schaberl (2022), we use the following final regressions:

 $LnP_{it} = \beta_0 + \beta_1 LnEPS_{it} + \beta_2 LnBV_{it} + \beta_3 POLCON_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 LNSIZE_{it} + \beta_7 LEV_{it} + \beta_8 MANOWN_{it} + \beta_9 INSTOWN_{it} + \beta_{10} LNAGE_{it} + \beta_{11}CG2004_{it} + \beta_{12}INDUSTRY_{it} + \beta_{13}POL^*EPS_{it} + \beta_{14}POL^*BV_{it} + \varepsilon_{it}$ (Equation 1)

 $LnP_{it} = \beta_0 + \beta_1 LnEPS_{it} + \beta_2 LnBV_{it} + \beta_3 POLCON_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 BODFRANCO*LnEPS_{it} + \beta_7 BODFRANCO*LnBV_{it} + \beta_8 BODINDO*LnEPS_{it} + \beta_9 BODINDO*LnBV_{it} + \beta_{10} POLCON*LnEPS_{it} + \beta_{11} POLCON*LnBV_{it} + \beta_{12} POLCON*LnEPS_{it} + \beta_{13} POLCON*LnBV_{it} + \beta_{14} POLCON*BODFRANCO_{it} + \beta_{12} POLCON*LnEPS_{it} + \beta_{13} POLCON*LnBV_{it} + \beta_{14} POLCON*BODFRANCO_{it} + \beta_{14} POLCON*LnEPS_{it} + \beta_{14} POLCON*BODFRANCO_{it} + \beta_{14$

(Equation 2)

 $\beta_{15}POLCON*BODINDO_{it} + \beta_{16}POLCON*BODFRANCO*LnEPS_{it} + \beta_{17}POLCON*BODFRANCO*LnBV_{it} + \beta_{18}POLCON*BODINDO*LnEPS_{it} + \beta_{19}POLCON*BODINDO*LnBV_{it} ----+ control variables + \varepsilon_{it}$

Independent variables

4.2.

The political connection, $POLCON_{it}$, will be measured using Faccio's (2006) definition of politically connected firms and its methodology. In this study, the political connection variable takes the value of 1 if the firm is considered a politically connected firm; otherwise, a value of θ -zero is granted. The methodology for identifying a politically connected firm is presented in Appendix B.

Board ethnic diversity has been measured based mainly on the ethnicity of the Mauritian population. The directors have been classified as Franco-Mauritian directors $(BODFRANCO_{ii})_{\tau}$ and Indo-Mauritian directors $(BODINDO_{ii})$. Franco-Mauritian directors are Mauritian whites, mostly descendants of French and British colonizers. In contrast, Indo-Mauritian directors originated from India and reached Mauritius as indentured laborers, merchants, and traders, and some as high-profile teachers and officers under British rule.

4.3. Control Variables

Following recommendations from the existing literature, control variables are exploited to manage the effect of firm-related variables: Size of the firm ($LNSIZE_{it}$), leverage of the firm (LEV_{it}), direct managerial ownership of the firm ($MANOWN_{it}$), institutional ownership of firm ($INSTOWN_{it}$), and lastly firm's age since incorporation ($LNAGE_{it}$). The firm's size is operationalized by the natural log transformation of total assets of firm *i* at

time *t* in millions of Mauritian Rupees, while leverage is operationalized by the natural logarithm of total debt divided by total assets. Direct managerial and institutional ownership are quantified by the direct managerial ownership's percentage of firm *i* and the ownership's percentage by the top 5 direct institutional shareholders of firm *i*. Finally, the age of firms are calculated by the natural logarithm of the number of years since incorporation.

Industry effects are also controlled by the industry dummies, categorized into eight industries based on Mauritius's Official Market, the Stock Exchange.⁵ Following the context of Mauritius, a dummy variable representing the year 2004 (*CG2004*) has been included to capture the effect of the enactment and implementation of the Code of Corporate Governance. Please refer to Appendix A for the variables' definitions.

(Appendix A here)

5. Results

5.1. Descriptive Statistics

Table 2 shows the descriptive statistics relating to this research. Panel A shows the descriptive statistics for the different variables included in Ohlson's value relevance model: the share price (*PRICE*_{*it*}), the Earnings per share (*EPS*_{*it*}), and the Book Value of shares in Mauritian Rupees of SEM-listed firms for the period 2001 – 2016. Panel B of Table 2 reports the descriptive statistics of the independent variables: Political connection and ethnicity. Political connection (*POLCON*_{*it*}) takes the value of 1 if the firm is politically connected; therefore, 36.6% of the sample have some political connections as per the descriptive statistics. Appendix B explains our methodology for identifying the politically connected firms in Mauritius.

(Appendix B here)

Board diversity shows that, on average, 68% of directors are white, while 32% of directors are non-white. Ethnicity is further analyzed by the percentage of Franco-Mauritian directors ($BODFRANCO_{it}$) and Indo-Mauritian directors ($BODINDO_{it}$). The percentage average percentages of $BODFRANCO_{it5}$ and $BODINDO_{it}$ is and 7.4%, respectively. Boards of directors are mostly composed of Franco-Mauritian directors. This is explained by Mauritius being an ex-British and ex-French colony.

(Table 2 here)

5.2. Correlations

Table 3 presents the Pearson and Spearman correlations for the variables included in Ohlson's model of value relevance, the experimental variables applicable to the context of Mauritius, and the control variables. They both yield the samesimilar results; however, the results from the Spearman correlations are more significant. The significant correlations grant initial support to support the hypotheses that political connections and the ethnic diversity of the board of directors influence the dependent variable. Both Pearson and Spearman correlations indicate that it will be relevant to test the hypotheses on the value relevance of accounting information.

(Table 3 here)

5.3. Univariate

5.3.1. Political connections

Table 4 exhibits the results from the univariate analysis for the test variables relating to politically connected and politically unconnected firms. Panel A of Table 4 shows that the share price varies for both groups; politically connected firms demonstrate higher share prices (*PRICE*_{ii}) than non-politically connected firms (Rs 67.121 as opposed to Rs 58.589). The same is observed for the log variable $LnPRICE_{ii}$. The earnings per share (*EPS*_{ii}) and the book value of equity (*BV*_{ii}) and their associated log variables are higher for politically connected firms than politically unconnected firms, which suggests that the value relevance of accounting information is potentially different for both groups. Furthermore, politically connected firms are sizeable, more leveraged, and have higher institutional ownership but less managerial ownership, as per panel C of Table 4.

(Table 4 here)

5.3.2. Ethnicity

We extend the univariate analysis by separating the sample based on the median values of the ethnic groups: $BODFRANCO_{it}$ and $BODINDO_{it}$ as tabulated in Tables 5 and 6, respectively. Firms that have more than the median value of $BODFRANCO_{it}$ recorded significantly higher $LnPRICE_{it}$, higher earnings per share $(LnEPS_{it})$ and book value $(LnBV_{it})$, and fewer firms considered as politically connected $(POLCON_{it})$. Not surprisingly, firms above the median value of $BODFRANCO_{it}$ have significantly fewer $BODINDO_{it}$ directors. Furthermore, these firms are significantly smaller, have higher managerial ownership $(MANOWN_{it})$, Mann-Whitney only), and are significantly older than firms with less than the median value of BODFRANCO directors.

(Table 5 here)

We perform a similar test for *BODINDO*_{it}. We find no significant differences in *LnPRICE, LnEPS* and *LnBV* between the two groups of firms<u>between the two groups of</u> firms in *LnPRICE, LnEPS*, and *LnBV*. As expected, firms with more than the median value of *BODINDO*_{it} are likely to be connected. Further, firms with a higher *BODINDO*_{it} have many significantly lower *BODFRANCO*_{it} directors. Panel C of Table 6 presents the control variables' mean and median differences. Firms with more than the median value of *BODINDO*_{it} are significantly larger, have a significantly lower level of managerial ownership, and are younger firms.

These univariate analyses provide insight into the differences in firm characteristics for firms dominated by $BODFRANCO_{it}$ or $BODINDO_{it}$. A significant finding is that firms that $BODINDO_{it}$ dominates are more politically connected. In addition, the univariate analysis supports our argument that firms with a high level of $BODINDO_{it}$ directors are also politically connected.

(Table 6 here)

5.4. Multivariate

Table 7 presents the main regressions. Column 1 of Table 7 presents the baseline regression for Ohlson's model, and we find that $LnEPS_{it}$ and $LnBV_{it}$ are positively and significantly associated with $LnPRICE_{it}$, at the one percent level. Column 2 of Table 7 tabulates the regression to include the control variables. We find a positive and significant relationship between $POLCON_{it}$ and $PRICE_{it}$ ($\beta = 0.130$, t=1.976, p<0.1), suggesting that political connections positively affect share prices. We find a negative and significant

relationship between *BODFRANCO_{it}* and *PRICE_{it}* ($\beta = 0.309$, *t=-4.266*, *p*<0.01)₅ and no significant relationship for *BODINDO_{it}*. All the control variables are significantly associated with *PRICE_{it}*, except *LNSIZE_{it}* and *LEV_{it}*.

Column 3 of Table 7 presents the regression to test the value relevance of political connections (*POLCON_{iv}*). We find the coefficient for *POLCON*BV_{it}* is negative and significant ($\beta = -0.180$, t = -2.688, p < 0.05), and this signals that political connections reduce the value relevance of book value (BV). However, we could not find any evidence for *POLCON*EPS_{it}*. All control variables remain similar to column 2 of Table 7. The results overall support the notion that political connections decrease earnings quality. In addition, the results are similar to other political connection studies (please see <u>Gul, 2006;</u> Fung *et al.*, 2015; <u>Gul 2006</u>).

(Table 7 here)

5.5. Testing for endogeneity

Political connections have been proven to be endogenous by several studies (Abdul Wahab *et al.*, 2011; Domadenik *et al.*, 2016). Maddala (1991) argues that selection bias happens once observations are sorted into discrete groups in a non-randomly mannernon-randomly, leaving way to-for potential coefficient bias in ordinary least squares procedures. Indeed, political connection scores are not allocated randomly but based on the firm's relationship with the government over time.

A two-step procedure is suggested: the first stage involves identifying the endogenous independent variable and addressing the selection model by using a probit regression to compute the inverse Mills ratio ($IMILLS_{it}$), which will be, in the second

phase, added to the main regression. The following first-stage model is applied to determine the inverse Mills ratio (*IMILLS_{it}*):

 $POLCON_{it} = \beta_0 + \beta_1 BIG4_{it} + \beta_2 LnEPS_{it} + \beta_3 LnBV_{it} + \beta_4 BODFRANCO_{it} + \beta_5 BODINDO_{it} + \beta_6 LNSIZE_{it} + \beta_7 LEV_{it} + \beta_8 MANOWN_{it} + \beta_9 INSTOWN_{it} + \beta_{10} LNAGE_{it} + \beta_{11}CG2004_{it} + \beta_{12}INDUSTRY_{it}$ (Equation 5)

This regression equation differs slightly from the main regression equation as an exclusion restriction – BIG4 has been added to the first stage of probit regression. The first added variable, BIG4 is a dummy variable that holds the value of 1 if the auditor is one of the Big 4 international auditing firms; 0 otherwise. Many studies have incorporated the relationship between the choice of auditors and political connections. Guedhami *et al.* (2014) investigate the relationship between political connections and auditor choice and find that connected firms are more likely to appoint a Big 4 auditor, as the insiders in these firms are eager to improve accounting transparency and convince outside investors that they are not misappropriating resources. Liu *et al.* (2017) examined the relationship between political connected firms and found that politically connected firms hire auditors of lower quality. Cheng *et al.* (2015) reached the same conclusion.

However, several recent studies have found that political connections adversely determine the likelihood of engaging quality auditors. Cheng *et al.* (2015), Habib *et al.* (2017), and Liu *et al.* (2017) argue that the likelihood of engaging quality auditors by politically connected firms is driven by managerial incentives to distort numbers, expropriate assets, and mask related-party transactions. These studies find results consistent with their arguments. Therefore, it is argued that a relationship between political connections and the choice of auditors exists.

Column 1 of Table 8 presents the probit regression when we include $BIG4_{it}$ as a determinant for *POLCON*. The coefficient for $BIG4_{it}$ and the *IMILLS_{it}* variable in columns 2 and 3 of Table 8 are insignificant, suggesting that the regressions do not suffer from selection bias. After controlling for selection bias, the results remain qualitatively similar to Table 7.

(Table 8 here)

5.6. *Political connections, board diversity and value relevance*

Table 9 presents the regression results when we test the impact of the ethnic groups on the value relevance of political connections. Column 2 of Table 9 presents shows the preliminary test for the value relevance for *BODFRANCO_{it}* and *BODINDO_{it}*. We find *BODFRANCO_{it}* and *BODINDO_{it}* reduce the value relevance of *EPS_{it}* on *PRICE_{it}*. However, we find contrasting results for *BV_{it}*.

Column 3 of Table 9 tabulates the regression for *BODFRANCO*_{it}. We find t<u>T</u>he coefficient for *POLCON*BODFRANCO*EPS*_{it} is positive and significant (β =0.639, *t*=3.434, *p*<0.01). This provides evidence that *BODFRANCO*_{it} mitigates the negative impact of political connections on the value relevance of *EPS*_{it}. Column 4 of Table 9 presents the result for *BODINDO*_{it}. The interaction term *POLCON*BODINDO*EPS*_{it} is negative and significant (β =-1.251, t=-3.320, *p*<0.01). The results support our arguments that *BODFRANCO*_{it} directors are experienced and probably more trustworthy to various stakeholders and mitigate the negative perception of political connections.

Furthermore, we find that the negative impact of political connections on value relevance worsens with firms with a high level of $BODINDO_{it}$ directors. The results are not surprising since $BODINDO_{it}$ directors are often linked to the government, and their

presence on the board could be largely ceremonial or an indicator of trying to tap into government grants. This indicator alone is usually perceived negatively since it signals the firm's inability to tap into the external market.

(Table 9 here)

6. Additional Analyses

We extend the test by separating the sample based on the median values of the ethnic groups; $BODFRANCO_{it}$ and $BODINDO_{it}$. The premise of our test is to investigate the collective impact of ethnic groups in mitigating the value relevance of political connections. The directors from various ethnic groups are expected to interact on the board. We argue that firms with Franco-Mauritian directors' domination will weaken the higher negative impact caused by the presence of Indo-Mauritian directors. In contrast, we argue that the positive impact of Franco-Mauritian directors will be weakened in firms that Indo-Mauritians dominate. In order tTo investigate these conjectures, we re-run the regressions presented in Table 9 by separating the samples according to the median values of $BODFRANCO_{it}$ and $BODINDO_{it}$, with minor adjustments. Table 10 presents the results for the impact of $BODFRANCO_{it}$ and $BODINDO_{it}$ and $BODINDO_{it}$ directors, respectively.

Columns 1 and 2 present the regressions when we separate the sample based on the median values of $BODFRANCO_{it}$. The results are similar to the earlier results ose tabulated in Tables 7 and 9. We find the interactions $POLCON*BODINDO*LnEPS_{it}$ and $POLCON*BODINDO*LnBV_{it}$ are not significant.

Columns 3 and 4 of Table 10 tabulate the regression when we split the sample based on the median value of $BODINDO_{it}$. We find that the results for the interactions $POLCON*BODFRANCO*EPS_{it}$ and $POLCON*BODFRANCO*BV_{it}$ are not significant,

suggesting that the impact of the presence of *BODINDO*_{it} is relatively minimal or non-existence.

(Table 10 here)

6.1. Elections

During the sample period, Mauritius experienced three elections in 2005, 2010, and 2014. We assigned dummy variables to control for each of the elections. The untabulated results find that the election years are significantly related to *PRICE*. The main test variables' results remain qualitatively similar to Table 7.

7. Conclusion

This study has two central objectives. The first and baseline objective is establishing the relationship between political connections and value relevance in MauritiusMauritius's political connections and value relevance. We are motivated to conduct such an examination as there is limited evidence of Mauritius's political connections. The political landscape in Mauritius is very interesting and is seen as a country dominated by two prominent families. Based on our analysis of 541 firm-year observations for 2001-2016 and the operationalization of political connections based on Faccio's (2006) seminal work, we find that political connections mitigate the value relevance of earnings and book value. The results are consistent with studies (e.g. Chen *et al.*, 2010; Chaney *et al.*, 2011; Chen *et al.*, 2010) that examine the impact of political connections on earnings quality. Further, the results remain similar after we perform a selection test.

The second and primary research objective is to examine whether board ethnicity mitigates the negative impact of political connections on value relevance. Mauritius is blessed with diverse ethnic groups working collectively since gaining independence. We conjecture that the ethnic groups in Mauritius represent the political landscape. We chose two contrasting ethnic groups: Franco-Mauritians, who <u>are dominatingdominate</u> the capital market but are a minority in the population, and Indo-Mauritians, who control the government and are a majority.

We find contrasting results based on the two ethnic groups, Franco-Mauritians, and Indo-Mauritians. We find firms that have Franco-Mauritians as directors mitigate the negative impact of political connections on the value relevance of earnings and book value. In contrast, we find that Indo-Mauritian directors worsen the negative impact of political connections on value relevance. The contrasting findings highlight the unique institutional settings in Mauritius. The results add to the extant literature, especially on the role of ethnicity in the involvement of the capital market via political connections. The findings of this study should act as a catalyst for further research on political connections in this region, which is often deemed 'sensitive' for accounting researchers-

¹ The Court found him not guilty in November 2019 as the pieces of evidence presented by the prosecuting parties were vague and uncertain and therefore failed the test of certainty required under the provisions of the Law (Police V N. Ramgoolam, 2019).

² Mauritius had a Franco-Mauritian Prime Minister; Paul Berenger, between 2003-2005.

³ Mauritius has an anti-corruption score of 73.1 out of 100 in the Mo Ibrahim Index in 2020 and ranked second out of 54 in Africa (Mo Ibrahim Foundation, 2021

⁴ It is a customized version of the Westminster majoritarian model whereby the constitution appoints for a Head of State or President, Vice–President, Prime Minister or Head of Government, Deputy Prime Minister(s), Members of Parliament, Parliamentary opposition with an appointed Leader of the opposition and many parliamentary secretaries.

⁵ The industries are Banks and Insurance; Commerce; Manufacturing; Investments; Leisure and Hotels; Property Development; Sugar Industry and Transport.



References

- Aapravasi Ghat Trust Fund- (2015)-, "Factsheet Aapravasi Ghat World Heritage Site", available
 at: https://aapravasi.govmu.org/aapravasi/wp-content/uploads/2020/09/Factsheet-Aapravasi-Ghat-World-Heritage-Site.pdf

 (accessed
 27
 September
 27, 2021, from)

 https://aapravasi.govmu.org/aapravasi/wp-content/uploads/2020/09/Factsheet-Aapravas
- Abdul Wahab, E.-A., Zain, M.-M., and James, K. (2011), "Political connections, corporate governance and audit fees in Malaysia", Managerial Auditing Journal, Vol. 26 No. (5), pp. 393–418. https://doi.org/10.1108/0268690111129562

Aapravasi-Ghat-World-Heritage-Site.pdf

- Abdul Wahab, E.-A., Pitchay, A.-A., and Ali, R. (2015), "Culture, corporate governance and analysts forecast in Malaysia", Asian Review of Accounting, Vol_23 (No. 3), pp. 232-255. https://doi.org/10.1108/ARA-03-2014-0033
- Abdul Wahab, E.-A., Madah Marzuki, M., Jaafar, S.-B., and Masron, T.-A. (2018), <u>"Board diversity and total directors' remuneration: evidence from an emerging</u> <u>market"</u>, Pacific Accounting Review, Vol. -30 No. (2), pp. – 243–272. https://doi.org/10.1108/PAR-02-2016-0021
- Abdullah, S.-N. (2013)₂- "*Causes of gender diversity in Malaysian firms*"-, Journal of Management and Governance, Vol. 18 No. (4), pp. 1137-1159. https://doi.org/10.1007/s10997-013-9279-0
- Adedeji, S.-B. and, Kajola, S.-O. (1998), Cost Accounting Concept, Analysis and Use for Decision Making. Centre for Sandwich Programme (CESAP).
- Adhikari, A., Derashid, C., and Zhang, H. (2006), "Public Policy, Political Connections, and Effective Tax Rates: Longitudinal Evidence from Malaysia", Journal of Accounting and Public Policy, Vol. 25, pp. 574–95. https://doi.org/10.1016/j.jaccpubpol.2006.07.001
- Agarwal, A. and, Knoeber, C.-R. (2001), "Do some outside directors play a political role?", Journal of Law and Economics, Vol. 44 No. (1), pp. 179–198. https://doi.org/10.1086/320271
- Alali, F. A., Foote, P. S. (2012). The Value Relevance of International Financial Reporting Standards: Empirical Evidence in an Emerging Market. International Journal of Accounting, University of Illinois, 47(1), 85–108. <u>https://doi.org/10.1016/j.intacc.2011.12.005</u>
- Arfken, D.-E., Bellar, S.-L.<u>and</u>, Helms, M. M. (2004), "*"The ultimate glass ceiling revisited: The presence of women on corporate boards"*, Journal of Business Ethics, <u>Vol. 50 No. (2)</u>, pp. 177–186. https://doi.org/10.1023/B:BUSI.0000022125.95758.98
- Badu, B. and, Appiah, K.-O. (2018), "Value relevance of accounting information: an emerging country perspective", Journal of Accounting and Organizational

Change, Vol. 14<u>No. (4), pp.</u> 473–491. <u>https://doi.org/10.1108/JAOC-07-2017-0064</u>

- Ball, R. and, Brown, P. (1968), "An Empirical Evaluation of Accounting Income Numbers", Journal of Accounting Research, Vol. 6 No.(2), pp. 159–178. https://doi.org/10.2307/2490232
- Ball, R., Robin, A. and, Wu, J. (2003). "Incentives versus standards: Properties of accounting income in four East Asian countries". Journal of Accounting and Economics, Vol. 36, pp. 235–270. https://doi.org/10.1016/j.jacceco.2003.10.003
- Bao, B., Chow, L. (1999). The Usefulness of Earning and Book Value for Equity Valuation in Emerging Capital Markets: Evidence from Listed Companies in the Peoples Republic of China. Journal of International Financial Management and Accounting, 10(2), 85–104. https://doi.org/10.1111/1467-646X.00045
- Baranchuk, N. and, Dybvig, P. (2009), "Consensus in diverse corporate boards", Review of Financial Studies, Vol. 22, pp. 715–747. https://doi.org/10.1093/rfs/hhn052
- Bart, C. and, McQueen, G. (2013), "*Why women make better directors?*", International Journal of Business Governance and Ethics, Vol. 8, pp. 93–99.
- Barth, M.-E., Beaver, W.-H. and, Landsman, W.-R. (2001), "*"The Relevance of the Value Relevance Literature For Financial Accounting Standard Setting: another view"*, Journal of Accounting and Economics, Vol. 31 No. (1), pp. 77–104. https://doi.org/10.1016/S0165-4101(01)00019-2
- Batta, G., Sucre Heredia, R. and, Weidenmier, M. (2014) "Political Connections and Accounting Quality under High Expropriation Risk" - European Accounting Review, Vol. 23 No. (4), pp. 485–517. https://doi.org/10.1080/09638180.2014.906316
- Beaver, W.-H. (1968)₃- "*The Information Content of Annual Announcements Earnings*-", Journal of Accounting Research, <u>Vol.</u>6, <u>pp.</u>67–92. https://doi.org/10.2307/2490070
- Beaver, W.-H. (2002)_a- "Perspectives on recent capital market research"-a Accounting Review, Vol. 77(No. 2), pp. 453–474. https://doi.org/10.2308/accr.2002.77.2.453
- Beaver, W.-H., Lambert, R. and, Morse, D. (1980), "*"The information content of security prices. Journal of Accounting and Economics"*, <u>Vol. 2 No. (1), pp.</u> 3-28. https://doi.org/10.1016/0165-4101(80)90013-0
- Beebeejaun, A.<u>and</u>, Koobloll, J. (2018)_s- "An assessment whether shareholder activism can be a corporate governance driver in the case of Mauritius-", International Journal of Law and Management, <u>Vol. 60(No. 6)</u>, pp. 1313–1337. https://doi.org/10.1108/IJLMA-07-2017-0154
- Beisland, L. A., Hamberg, M. (2013). Earnings sustainability, economic conditions and the value relevance of accounting information. Scandinavian Journal of Management 29(3), 314–324. <u>https://doi.org/10.1016/j.scaman.2013.02.001</u>

Beisland, L., Hamberg, A.-M. and, Novak, J. (2010)₃, "The Value Relevance Across Industries: What Happened to the New Economy?", Financial Management <u>Association.</u>

http://www.fma.org/Prague/Papers/Value_Relevance_across_Industries.pdf.

- Bernard, V. (1995). The Feltham-Ohlson framework: implications for empiricists. Contemporary Accounting Research, 11(2), 733-747. https://doi.org/10.1111/j.1911-3846.1995.tb00463.x
- Bertrand, M., Kramarz, F., Schoar, A. and, Thesmar, D. (2004)₃, "Politically Connected CEOs and Corporate Outcomes: Evidence from France"₃, wWorking paper, University of Chicago.
- Bilimoria, D. and Piderit, S. (1994) = "Board committee membership: effects of sex-based bias", Academy of Management Journal, Vol. 37, pp. 1453–1477. https://doi.org/10.2307/256795
- Boonlert-U-Thai, K. and, & Schaberl, P. (2022)₃, "Value relevance of book values, earnings, and future earnings: Evidence by time, life cycle stage, and market uncertainty". Asian Review of Accounting, ahead-of-print.
- Boubakri, N., Guedhami, O., Mishra, D. and, Saffar, W. (2012), "Political connections and the cost of equity capital", Journal of Corporate Finance, Vol. 18 No (3), pp. 541–559. https://doi.org/10.1016/j.jcorpfin.2012.02.005
- Brautigam, D. (withand Diolle, T₂). (2009)₃- "Coalitions, capitalists, and credibility: Overcoming the crisis of confidence at independence in Mauritius",--(_Rresearch Ppaper 04)-, Developmental Leadership Program. available at: <u>https://www.dlprog.org/publications/research-papers/coalitions-capitalists-andcredibility-overcoming-the-crisis-of-confidence-at-independence-in-mauritius (accessed 23 March 2018)<u>http://www.dlprog.</u> org/ftp/info/Public%20Folder/1%20Research%20Papers/Coalitions,%20 Capitalists%20and%20Credibility.pdf</u>
- Bunwaree, S.<u>and</u>, Kasenally, R. (2005)_a, *Political Parties and Democracy in Mauritius*, (RResearch RReport No 19,), EISA, South Africa.
- Bushman, R., Chen, Q., Engel, E. and ,-Smith, A. (2004)_a- "Financial accounting information, organizational complexity and corporate governance systems"_{-a} Journal of Accounting and Economics, <u>Vol.</u> 37, <u>pp.</u> 167-201. <u>https://doi.org/10.1016/j.jacceco.2003.09.005</u>
- Carroll, B.-W. and, Carroll, T. (2000), "-Trouble in paradise: Ethnic conflict in Mauritius", Journal of Commonwealth & Comparative Politics, Vol. 38(No, 2), pp. 25–50. https://doi.org/10.1080/14662040008447817
- Carter, D.-A., Souza, F.-D., Simkins, B.-J. and, Simpson, W.-G. (2010), "*"The Gender and Ethnic Diversity of US Boards and Board Committees and Firm*", Corporate Governnace: An International Review, <u>Vol.</u> 18 <u>No.</u> (5), pp. 396–414. https://doi.org/10.1111/j.1467-8683.2010.00809.x

Chaney, P.-K., Faccio, M. and, Parsley, D. (2011),- "The quality of accounting

information in politically connected firms["]₂, Journal of Accounting and Economics, Vol. 51, pp. 58–76. <u>https://doi.org/10.1016/j.jacceco.2010.07.003</u>

- Chen, C.-J.-P., Ding, Y. and, Kim, C. (2010), "*High-Level Politically Connected Firms, Corruption, and Analyst Forecast Accuracy Around the World*", Journal of International Business Studies, Vol. 41, pp. 1505 1524. https://doi.org/10.1057/jibs.2010.27
- Cheng, C.-L., Hsu, C.-S. and, Kung, F.-H. (2015)_s- "Political connections, managerial incentives and auditor choice: evidence from China", Pacific Accounting Review, Vol. 27 (No. 4), pp. 441–465. https://doi.org/10.1108/PAR-10-2013-0096
- Chiriyankandath, J. (2009), "-Global indianIndian diasporas: Exploring trajectories of migration and theory".- Journal of Global History, Vol. 4 No. (3), pp. 500-502. https://doi.org/10.1017/S1740022809990209
- CIA (2021), "The world factbook", available at: https://www.cia.gov/the-world-factbook/countries/mauritius/#people-and-society (accessed 6 December 2021)
- Civilize, S., Wongchoti, U. and , M. Young, M. (2015), "Political Connection and Stock Returns: A Longitudinal Study.", The Financial Review, Vol. 50 No.(1), pp. 89– 119. https://doi.org/10.1111/fire.12061
- Collins, D.-W., Maydew, E.-L. and, Weiss, I.-S. (1997), "Changes in the value-relevance of earnings and book values over the past forty years", Journal of Accounting and Economics, Vol. 24 No. (1), pp. 39–67. https://doi.org/10.1016/S0165-4101(97)00015-3
- Dale-Olsen, H., Schone, P.<u>and</u>, Verner, M. (2013)₃, <u>"Diversity among Norwegian</u> boardsnof directors: Does a quota for women improve performance" Feminist Economics, <u>Vol. 19</u><u>No.(4)</u>, <u>pp.</u> 110-135. <u>https://doi.org/10.1080/13545701.2013.830188</u>
- Darga, L. A.<u>and</u>; Joomun, G.–D. (2005), <u>"</u>-*Mauritius*", South African Journal of International Affairs, <u>Vol.</u> <u>12</u> <u>No.</u> (1), <u>pp.</u> 95–110. <u>https://doi.org/10.1080/10220460509556751</u>
- Diftar, D. and, Elkalla, T. (2019), "*The value relevance of accounting information in the MENA region*", Journal of Financial Reporting and Accounting, Vol. 17(No. 3), pp. 519-539. https://doi.org/10.1108/JFRA-09-2018-0079
- Dinç, I.-S. (2005)_a- "Politicians and banks: Political influences on government-owned banks in emerging markets" Journal of Financial Economics, <u>Vol.</u> 77<u>No(.</u>2), pp. 453–479. <u>https://doi.org/10.1016/j.jfineco.2004.06.011</u>
- Distefano, J.-J.<u>and</u>, Maznevski, M.-L. (2000)₇₂ "Creating value with diverse teams in global management", Organizational Dynamics, Vol. 29 No. (1), pp. 45–63. https://doi.org/10.1016/S0090-2616(00)00012-7
- Domadenik, P., Prašnikar, J. and, Svejnar, J. (2016) "Political Connectedness, Corporate Governance, and Firm Performance" - Journal of Business Ethics,

Vol. 139 No. (2), pp. 411-428. https://doi.org/10.1007/s10551-015-2675-4

- Faccio, M. (2006)₇, <u>"Politically connected firms"</u>, The American Economic Review, <u>Vol.</u> 96<u>No. (1)</u>, pp. 369–386. https://doi.org/10.1257/000282806776157704
- Faccio, M., Masulis, R.-W. and, McConnell, J.-J. (2006) "Political Contributions and Corporate Bailouts-", The Journal of Finance, Vol. 61(<u>No. 6</u>), pp. 2597–2635. https://doi.org/10.1111/j.1540-6261.2006.01000.x
- Fan, J.-P.-H., Wong, T.-J. adnd, Zhang, T. (2007), "Politically connected CEOs, corporate governance, and Post-IPO performance of China's newly partially privatized firms", Journal of Financial Economics, Vol. 84 No. (2), pp. 330–357. https://doi.org/10.1016/j.jfineco.2006.03.008
- Fisman, R. (2001)₂, "Estimating the Value of Political Connections", The American Economic Review, Vol. 91 No. (4), pp. 1095–1102. https://doi.org/10.1257/aer.91.4.1095
- Francis, J. and, Schipper, K. (1999), "Have Financial Statements Lost Their Relevance?", Journal of Accounting Research, Vol. 37 No. (2), pp. 319–352. https://doi.org/10.2307/2491412
- Fung, Y.-K., Gul, F.-A. and, S. Radhakrishnan. (2015), "Corporate politically connections and the 2008 Malaysian election", Accounting Organization and Society, Vol. 43 No. (2), pp. 67-86. https://doi.org/10.1016/j.aos.2015.04.001
- Gist, W.-E. and, Abdul Wahab, E.-A. (2021), "Political Patronage, Audit Quality and the Properties of Analysts' Earnings Forecasts in Malaysia", Journal of International Accounting Research, <u>Vol.</u> 20<u>No</u>(1), pp. 49–78. https://doi.org/10.2308/JIAR-19-077
- Glasscock, R., Korenok, O. and, & Dorminey, J. (2022)₃, "Spurious Correlation Due to Scaling"_3, Journal of Accounting, Auditing & Finance, https://doi.org/10.1177/0148558X2110637040148558X211063704.
- Goodstein, J., Gautam, K.<u>, and</u> Boeker, W. (1994)<u>,</u> "*The effects of board size and diversity on strategic change*", Strategic Management Journal, <u>Vol.</u> 15 <u>No.</u>(-3), <u>pp.</u> 241-250. https://doi.org/10.1002/smj.4250150305
- Guedhami, O., Pittman, J.-A.<u>and</u>, Saffar, W. (2014)<u>5</u>, "Auditor Choice in Politically Connected Firms"<u>5</u> Journal of Accounting Research, <u>Vol.</u> 52(<u>No.</u> 1), pp. 107-162. https://doi.org/10.1111/1475-679X.12032
- Gul, F.-A. (2006)₂, <u>"</u>*Auditors' response to political connection and cronyism in Malaysia*"₂, Journal of Accounting Research, <u>Vol. 44 No. (5)</u>, pp. 931-963. https://doi.org/10.1111/j.1475-679X.2006.00220.x
- Habib, A., Muhammadi, A.-H. and, Jiang, H. (2017), "Political Connections and Related Party Transactions: Evidence from Indonesia", International Journal of Accounting, <u>Vol.</u> 52 <u>No.</u> (1), <u>pp.</u> 45–63. <u>https://doi.org/10.1016/j.intacc.2017.01.004</u>

- Hanousek, J., Kocenda, E. and, Svejnar, J. (2007)₃- "Origin and concentration: Corporate ownership, control and perfonnance in firms after privatization". Economics of Transition, Vol. 15 No. (1), pp. -1-31. https://doi.org/10.1111/j.1468-0351.2007.00278.x
- Hillman, A.-J., Shropshire, C.<u>and</u>, Cannella, A.-A. (2007), "Organizational predictors of women on corporate boards", - Academy of Management Journal, <u>Vol. 50 No.</u> (4), pp. 941–952. htts://doi.org/ 10.5465/AMJ.2007.26279222
- Holmberg, L. (1962).- "Mauritius. A study in disaster".- Economy and History, Vol. 5 No. (1), pp. 3-29. https://doi.org/10.1080/00708852.1962.10418985
- Johnston, D. and, Malina, M.-A. (2008), "Managing sexual orientation diversity: The impact on firm value", Group and Organisation Management, Vol. 33, pp. 602– 625. https://doi.org/10.1177/1059601108321833
- Johnson, S. and, Mitton, T. (2003), "Cronyism, and capital controls: Evidence from Malaysia", Journal of Financial Economics, Vol. 67, pp. 351-382. https://doi.org/10.1016/S0304-405X(02)00255-6
- Jones, S.-L., Megginson, W.-L., Nash, R.-C. and, Netter, J.-M. (1999), "Share issue privatizations as financial means to political and economic ends", Journal of Financial Economics, Vol. 53 No. (2), pp. 217-253. https://doi.org/10.1016/S0304-405X(99)00021-5
- Khwaja, A.-I.<u>and</u>, Mian, A. (2005)₂, "Do Lenders Favor Politically Connected Firms? *Rent Provision in an Emerging Financial Market*", The Quarterly Journal of Economics, <u>Vol.</u> 120 <u>No.</u> (4), <u>pp.</u> 1371–1411. https://doi.org/10.1162/003355305775097524
- Kim, M. and -Kross, W. (2005) "The ability of earnings to predict future operating cash flows has been increasing - Not decreasing". Journal of Accounting Research, Vol. 43 No. (5), pp. 753–780. https://doi.org/10.1111/j.1475-679X.2005.00189.x
- Kim, Y., Li, S., Pan, C. and, Zuo, L. (2013), "The role of accounting conservatism in the equity market: evidence from seasoned equity offerings", The Accounting Review, Vol. 88 No. (4), pp. 1327–1356. https://doi.org/ 10.2308/accr-50420
- Kothari, S. (2001)₂- <u>"Capital markets research in accounting"</u>_{-z} Journal of Accounting and Economics, <u>Vol. 31 No. (1-3)</u>, pp. 105–231. <u>https://doi.org/10.1016/S0165-4101(01)00030-1</u>
- Lange, M. (2003)₃- <u>"Embedding the colonial state: A comparative-historical analysis of</u> state building and broad-based development in Mauritius<u>"</u>, Social Science History, <u>Vol.</u>27, <u>pp.</u>397-423. https://doi.org/10.1017/S014555320001258X
- Leuz, C. and, Oberholzer-Gee, F. (2006), "Political relationships, global financing, and corporate transparency: Evidence from Indonesia", Journal of Financial Economics, <u>Vol.</u> 81 <u>No.</u> (2), <u>pp.</u> 411–439. <u>https://doi.org/10.1016/j.jfineco.2005.06.006</u>
- Li, H., Meng, L., Wang, Q. and, Zhou, L. A. (2008), "Political connections, financing

- Liang, L.W., Lin, T.J., Yu, H.F. and Li, Y. W. (202149)₃₇ "The Impact of Political Connection and Board Diversity on Company Performance: An Example of Chinese-Listed CompaniesEvidence from China", Managerial and Decision Economics, Vol. 43 No. 6, pp. 2347-2357₇. https://doiorg.ezproxy.mdx.ac.uk/10.1002/mde.3529
- Lincoln, A. and, Adedoyin, O. (2012), Corporate governance and gender diversity in Nigerian boardrooms", World Academy of Science, Engineering and Technology, Vol. 71, pp. 1853–1859. https://ssrn.com/abstract=2201160
- Lipton, M.<u>and</u>, Lorsch, J. (1992)₁, <u>"</u>*A modest proposal for improved corporate governance*"_., Business Lawyer, <u>Vol.</u> 48, <u>pp.</u> 59–77. https://heinonline.org/HOL/P?h=hein.journals/busl48&i=95.
- Liu, Y., Li, X., Zeng, H. and, An, Y. (2017), "Political connections, auditor choice and corporate accounting transparency: evidence from private sector firms in China", - Accounting and Finance, Vol. 57 No. (4), pp. 1071–1099. https://doi.org/10.1111/acfi.12207
- Liu, J. and, Liu, C. (2007)₃- "Value relevance of accounting information in different stock market segments: The case of Chinese A-, B- and H-shares", --Journal of International Accounting Research, Vol. 6, pp. 55–81. https://doi.org/10.2308/jiar.2007.6.2.55
- Lubberink, M.-J.-P., and Willett, R.-J. (2021),- "Specification of the Market-Accounting Relation",- Available at SSRN 3976635.
- Luckerath-Rovers, M. (2013)₂₇ "Women on boards and firm performance"₂₇ Journal of Management and Governance, <u>Vol.</u> 17, <u>pp.</u> 491–508. https://doi.org/10.1007/s10997-011-9186-1
- Maddala, G. (1991), *"A perspective on the use of limited-dependent and qualitative variables models in accounting research"*, The Accounting Review, Vol. 66, No. (4), pp. 788–807.
- Mahadeo, J.-D., Soobaroyen, T.<u>and</u>, Hanuman, V.-O. (2012)_a, "Board Composition and Financial Performance: Uncovering the Effects of Diversity in an Emerging Economy"<u></u>, Journal of Business Ethics, <u>Vol.</u> 105 <u>No.</u> (3), pp. 375–388. https://doi.org/10.1007/s10551-011-0973-z
- Mahadeo, J.-D. and, Soobaroyen, T. (2013), "A Longitudinal Study of the Implementation of the Corporate Governance Code in a Developing Country: The Case of Mauritius", Business and Society, Vol. 55 No. (5), pp. 738–777. https://doi.org/10.1177/0007650313501838
- Mishra, R.-K. and ,-Jhunjhunwala, S. (2013), "Diversity—Toward an Effective Board",... R. K. Mishra & S. Jhunjhunwala (Eds.), *Diversity and the Effective Corporate Board*, (pp. 1–11). Academic Press, Cambridge, Massachussetts, pp. 1–11.

- Munisi, G. and ,-Randøy, T. (2013),- "Corporate governance and company performance across Sub-Saharan African countries",- Journal of Economics and Business, <u>Vol.</u> 70, pp. 92–110. <u>https://doi.org/10.1016/j.jeconbus.2013.08.003</u>
- Nee, V., Opper, S. and,-Wong, S. (2007)-, "Development state and corporate governance in China"_- Management and Organization Review, Vol. 3 No. (1), pp. 19-53. https://doi.org/10.1111/j.1740-8784.2007.00061.x
- Negakis, J. (2005)-, <u>"Accounting and capital markets research: A review"</u>, Managerial Finance, <u>Vol.</u>-31 <u>No.</u>(2), <u>pp.</u> 1-23.
- Ntim, C.-G. (2013):<u>"</u>-Board diversity and organizational valuation: unravelling the effects of ethnicity and gender"... Journal of Management and Governance, <u>Vol.</u> 19.<u>No. (1), pp.</u> 167–195. https://doi.org/10.1007/s10997-013-9283-4
- Ohlson, J.–A. (1995)<u>,</u> *"Earnings, book-values, and dividends in equity valuation"*, Contemporary Accounting Research, <u>Vol. 11 No. (11)</u>, <u>pp. 661–687</u>. https://doi.org/10.1111/j.1911-3846.1995.tb00461.x
- POLICE v N. RAMGOOLAM. (2019). The Intermediate Court of Mauritius.
- Reuters- (2016)_a- "Mauritius environment minister resigns over graft allegations". Available at: https://www.reuters.com/article/ozatp-uk-mauritius-corruptionidAFKCN0WP1LZ
- Riahi-Belkaoui, A. (2004)... "Politically-Connected Firms: Are They Connected To Earnings Opacity?"... Research in Accounting Regulation, Vol. 17 No.(1), pp. 25– 38. https://doi.org/10.1016/S1052-0457(04)17002-1
- Roberson, Q.-M. and, Park, H.-J. (2007), "*Examining the link between diversity and firm performance: The effects of diversity reputation and leader racial diversity*", Group and Organization Management, Vol. 32, pp. 548–568. https://doi.org/10.1177/1059601106291124
- Salverda, T. (2013)₃- "Balancing (re)distribution: Franco- Mauritians Landownership in the Maintenance of an Elite Position", Journal of Contemporary African Studies, Vol. 31 No. (3, pp. 503-521). https://doi.org/10.1080/02589001.2013.812455
- Salverda, T.₅ (2015)₂- "The Franco-Mauritian Elite: Power and Anxiety in the Face of Change"-₃ Berghahn Books, Incorporated.
- Simkins, B.-J. and, Carter, D.-A. (2003), "Corporate Governance, Board Diversity, and *Firm Value*", The Financial Review, <u>Vol.</u> 38, <u>pp.</u> 33–53. https://doi.org/10.1111/1540-6288.00034
- Sonnenfeld, J.-A. (2002), "--What makes great boards great", Harvard Business Review, Vol. 80 No. (9), pp. 106-113.
- Spilioti, S. N., Karathanassis, G. A. (2010). Comparison of the Ohlson and Feltham-Ohlson Models for Equity Valuation: Evidence from the British Telecommunications Sector. Working Paper. www.eefs.eu/conf/Warsaw/Paper/672.doc

- Srebrnik, H. (2002), ""Full of sound and fury': Three decades of parliamentary politics in Mauritius. Journal of Southern African Studies", Vol. 28 No. (2), pp. 277–290. https://doi.org/10.1080/03057070220140702
- Srinidhi, B., Gul, F.-A. and, Tsui. J. (2011)₃- "Female directors and earnings quality-", Contemporary Accounting Research, Vol. 28 No. (5), pp. 1610–1644. https://doi.org/10.1111/j.1911-3846.2011.01071.x
- Tantawy, S.M. and Moussa, T. (2023), "The effect of political connections on firms' auditor choice decisions and audit opinions: evidence from Egypt", Asian Review of Accounting, ahead-of-print. https://doi-org.ezproxy.mdx.ac.uk/10.1108/ARA-07-2022-0161
- Tee, C.-M. (2019)₃- <u>"Political connections, the cost of debt and board attributes: evidence</u> from Malaysia-<u>"</u>, Managerial Finance, <u>Vol. 45 No. (7)</u>, pp. 842–855. https://doi.org/10.1108/MF-04-2018-0179
- Triana, M.-C., Miller, T.-L.<u>and</u>, Trezebiatowski, T.-M. (2014)₂, "*The Double-Edged Nature of Board Gender Diversity: Diversity, Firm Performance, and the Power of Women Directors as Predictors of Strategic Change*", Organization Science, Vol. 5 No. (2), pp. 609–32. https://doi.org/10.1287/orsc.2013.0842
- Van Der Walt, N. and ,-Ingley, C. (2003), "Board dynamics and the influence of professional background, gender and ethnic diversity of directors", Corporate Governance: An International Review, Vol. 11, pp. 218–234. https://doi.org/10.1111/j.1467-8683.2005.00456.x
- Wan, J. (2015, March 17)₄- "Changing fortunes of Mauritius' ousted leaders"<u>*</u> *Al Jazeera* <u>17</u> <u>March. Available at:</u> <u>+</u> https://www.aljazeera.com/features/2015/3/17/changing-fortunes-of-mauritius-ousted-leaders
- Wellage, N.-H. and, Locke, S. (2013) <u>s</u>-<u>"</u>*Corporate governance, board diversity and firm financial performance: New evidence from Sri Lanka*"<u>s</u>-International Journal of Business Governance and Ethics, <u>Vol. 8</u>, <u>pp.</u>116–136.
- World Bank₋ (2010)_a, "Report on the observance of standards and codes Corporate Governance (Mauritius)".

APPENDIX A

Variable	Operational Definition	Source(s)		
Panel A: Depen	dent Variable			
LnPRICE	Log transformation of Share Price of firm <i>i</i> for the period <i>t</i> in Mauritian Rupees.	Stock Exchange of Mauritius		
Panel B: Indepe	ndent Variables			
LnEPS	Natural log transformation of Earnings per share of firm i for the period t in Mauritian Rupees.	DataStream, BvD's Osiris, annual reports		
LnBV	Natural log transformation of Book value per share of equity of firm <i>i</i> at time <i>t</i> in Mauritian Runees	DataStream, BvD's Osiris, annual reports		
POLCON	Political connection of firm <i>i</i> at time <i>t</i> .			
BODFRANCO	Franco-Mauritian directors on the board of directors of firm <i>i</i> .	Annual reports		
BODINDO	Indo-Mauritian directors from Indian descent on the board of directors of firm <i>i</i> .	Annual reports		
Panel C: Contro	l Variables			
SIZE	Total Assets of firm <i>i</i> at time <i>t</i> in millions of	DataStream, BvD's		
	Mauritian Rupees.	Osiris, annual reports		
LNSIZE	Natural log transformation of Total Assets of firm <i>i</i> at time <i>t</i> in millions of Mauritian Runees	DataStream, BvD's Osiris, annual reports		
LEV	Natural log transformation of Total Debt divided by Total Assets.	DataStream, BvD's Osiris, annual reports		
MANOWN	Percentage of direct managerial ownership of firm <i>i</i> .	DataStream, BvD's Osiris, annual reports		
INSTOWN	Percentage ownership by the top five direct institutional shareholders of firm <i>i</i> .	DataStream, BvD's Osiris, annual reports		
AGE	Natural log transformation of the sum of years since incorporation	Annual reports		
Panel D: Exclus	ion Restriction			
BIG4	An indicator variable that takes the value of 1 if the firm engaged an international Big 4 auditor	Annual reports		

APPENDIX B

Politically connected firms

Politically connected firms were identified using Faccio (2006) 's definition of politically connected firms and/or significant government ownership in the firm. Therefore, politically connected firms satisfy any of the following:

- (a) Had Member of Parliaments (MPs) on the board of directors,
- (b) Had senior executives closely related to a MP or government official,
- (c) Had significant ownership by government or government-linked organizations.

The data was collected through:

- (a) Connections identified by linking the biodata of MPs and Government officials from the government websites and the biodata of senior executives in annual reports,
- (b) Percentage of ownership of government and government-related entities in annual report and companies' websites
- (c) Review of newspapers, government publications and firms' publications to established connections
- (d) Review of biodata of board members in annual reports.

Due to the sensitive nature of political connections in Mauritius and the current political turmoil in the country, the names of the firms are not disclosed:

Firm A	1.	A board member was the lawyer of one of the Prime ministers of
		Mauritius. The connection was identified through the obituary of this
		person as he passed away in 2013.
	2.	A board member was the brother of an MP: The connection was
		established through the biodata of the MP on the government website and
		the biodata of the director.
	3.	A board member was the sister-in-law of the Minister of Education and
		MP, she was the CEO of a state-owned organization and director on
		several government boards. This connection was established through the
		newspapers, the biodata of the Minister on the government website and
		the websites of the different organizations, of which she is a director.
	4.	Two government entities were also considered as major shareholders for
		the period 2009 – 2016. Connections identified in the annual reports.

 Table 1: Sample Selection

Description	Sample Size
Firm-year Observations 2001-2016 ^a	688
Less Observations from Financial Companies	(48)
	640
Less Observations with missing financial data	(99)
Firm-year Observations for the final sample	541

^a The sample was developed starting with all audited firms listed on the Stock Exchange of Mauritius during the period. During the 2001-2016 period of this study, there were 43 firms listed on the main market of the Stock Exchange of Mauritius.

	Mean	Median	Maximum	Minimum	
Panel A: Dependent variable	le				
PRICE	62.371	40.275	515	0.000	
LnPRICE	3.526	3.696	6.244	-0.236	
Panel B: Independent varia	bles				
EPS	260.058	3.295	146372.5	-39.77	
LnEPS	1.111	1.401	11.894	-6.645	
BV	7.135	10.000	10.700	0.010	
LnBV	1.544	2.303	2.370	-4.605	
POLCON	0.366	0.000	1.000	0.000	
BODFRANCO	0.607	0.667	1.000	0.000	
BODINDO	0.174	0.100	1.000	0.000	
Panel C: Control variables					
SIZE (millions)	13068.601	3257.542	317704.8	38.895	
LNSIZE	21.032	21.502	26.484	11.905	
LEV	-1.0945	-0.676	0.0284	-6.308	
MANOWN	0.0280	0.0002	7.361568	0.000	
INSTOWN	0.513	0.596	0.910	0.000	
AGE	62.84	41	227	1.000	
LNAGE	3.841	3.714	5.425	0.000	

Table 2: Descriptive Statistics

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. The logged variables LnPRICE, LnEPS, LnBV, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of DODE and the statement of the directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation.

			1									
Correlations		1	2	2	4	5	6	7	8	9	10	11
PRICE	1		.675***	.487***	.014	.090**	085**	.171***	.313***	.091**	.123***	.475**
EPS	2	.549***		.363***	.026	.006	035	.166***	.270***	007	004	.337**
BV	3	.289***	.200***		.031	.111**	051	046	.217***	.082*	.075*	.131**
POLCON	4	.058	.046	.079*		245***	.309***	.122***	.144***	.134***	109**	027
BODFRANCO	5	.055	.015	.077*	239***		727***	143***	194***	006	.175***	.194**
BODINDO	6	130***	053	010	.395***	649***		.183***	.130***	026	252***	113**
LNSIZE	7	.014	.112***	121***	.101**	137***	.196***		.402***	073*	.096**	.111**
LEV	8	.240***	.088**	.258***	.053	.016	139***	.121***		.248***	.148***	.319**
INSTOWN	9	.086**	069	.107**	.148***	018	048	174***	.346***		207***	.080**
MANOWN	10	.065	.018	.038	062	.004	047	.032	.039	022		019
LNAGE	11	.397***	.266***	.165***	009	.261***	265***	013	.342***	.033	050	

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection.

BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation.

Significance at: *10, **5 and ***1 per cent levels.

4	
5	
6	
7	
8	
9	
1	ი
1	1
1	י ר
1	2
1	3 ⊿
1	4
1	5
1	6
1	7
1	8
1	9
2	0
2	1
2	2
2	3
2	4
2	5
2	6
2	7
2	, 0
2	0 0
2	9
3	1
3	1
3	2
3	3
3	4
3	5
3	6
3	7
3	8
3	9
4	0
4	1
4	2
Δ	2
1	⊿
1	-+ 5
4	2
4	0
4	/
4	8

Table 4: Univariate Analysis of Differences variables between Politically and Non-Politically connected Firms in Mauritius (2001-2016)

	(n=	=199)	(n=	=342)	t - Tests	Mann
	POL	POLCON =1		CON =0		Whitney
	Mean	Median	Mean	Median	p-Value	p-Value
Panel A: Dependent va	ıriable	<u> </u>				_
PRICE	67.121	37.700	58.589	40.650	0.183	0.747
LnPRICE	3.592	3.630	3.444	3.705	0.187	0.747
Panel B: Independent	variables					
EPS	5.401	3.330	4.492	2.975	0.300	0.588
LnEPS	1.252	1.401	0.965	1.340	0.054	0.171
BV	7.592	10.000	6.918	10.000	0.054	0.484
LnBV	1.584	2.303	1.540	2.303	0.694	0.484
BODFRANCO	0.512	0.556	0.665	0.750	0.000	0.000
BODINDO	0.278	0.222	0.112	0.091	0.000	0.000
Panel C: Control varia	ıbles					
LNSIZE	21.418	21.942	20.814	21.455	0.017	0.004
LEV	-0.999	-0.553	-1.137	-0.718	0.210	0.001
MANOWN	0.002	0.000	0.043	0.000	0.151	0.010
INSTOWN	0.561	0.602	0.490	0.562	0.001	0.002
LNAGE	3.845	3.689	3.860	3.738	0.828	0.526

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. The logged variables *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors for Indian descent on the board of directors.
SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of ifrect managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is the natural log transformation of the sum of years since incorporation. Significant p-values are bold.

Table 5: Univariate Analysis of Differences variables between Firms that are below/equal and above the median value of BODFRANCO (2001-2016)

	BODFRA	NCO	BODFR	ANCO		Mann-
	= <median< th=""><th>n (n=279)</th><th colspan="2">>median (n=262)</th><th>t-test</th><th>Whitney</th></median<>	n (n=279)	>median (n=262)		t-test	Whitney
	Mean	Median	Mean	Median	p-value	p-value
Panel A: Dependent Variable					•	•
PRICE	57.319	33.500	66.795	43.150	0.113	0.001
LnPRICE	3.312	3.497	3.312	3.765	0.000	0.001
Panel B: Independent Variables						
EPS	4.622	3.270	5.083	3.030	0.559	0.593
LnEPS	0.909	1.296	1.250	1.480	0.017	0.037
BV	6.691	10.000	7.683	10.000	0.003	0.000
LnBV	1.401	2.303	1.722	2.303	0.003	0.000
POLCON	0.484	0.000	0.240	0.000	(0.000)	
BODINDO	0.287	0.250	0.054	0.000	0.000	0.000
Panel C: Control variables						
LNSIZE	21.511	21.834	20.513	21.423	0.000	0.002
LEV	-1.055	-0.568	-1.126	-0.768	0.463	0.000
MANOWN	0.018	0.000	0.039	0.000	0.466	0.000
INSTOWN	0.518	0.596	0.513	0.573	0.793	0.379
LNAGE	2 752	2 680	3 066	3 761	0.001	0.008

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. The logged variables *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold.
Table 6: Univariate Analysis of Differences variables between Firms that are below/equal and above the median value of BODINDO (2001-2016)

13								
14	Y	BODINDO =	< median	BODI	NDO		Mann-	Ī
15		value (n	-282)	>median	(n=259)	t-test	Whitney	
16		Mean	Median	Mean	Median	n-value	n-value	
17	Panel 4: Dependent Variable	wiedn	wiculan	Ivicali	Wiedian	p-value	p-value	-
18	PRICE	60 887	13 650	53 221	20.000	0.007	0.007	
19		09.887	43.030	2 450	29.000	0.007	0.002	
20	LNPRICE	3.557	3.695	3.450	3.611	0.322	0.586	
21	Panel B: Independent Variables							
22	EPS	5.472	3.830	4.164	2.620	0.123	0.011	
22	LnEPS	1.065	1.391	1.070	1.353	0.974	0.967	
24	BV	7.449	10.000	6.869	10.000	0.106	0.008	
25	LnBV	1.569	2.302	1.545	2.303	0.827	0.903	
26	POLCON	0.262	0.000	0.479	0.000	(0.000)		
27	BODFRANCO	0.791	0.833	0.408	0.455	0.000	0.000	
28	Panel C: Control variables							
29	LNSIZE	20.690	21.479	21.395	21.575	0.004	0.067	
30	LEV	-1.124	-0.750	-1.052	-0.580	0.494	0.001	
31	MANOWN	0.042	0.001	0.013	0.000	0.263	0.000	
32	INSTOWN	0.520	0.584	0.511	0.596	0.644	0.873	
33	LNAGE	3.938	3.738	3.767	3.714	0.011	0.180	
34								

PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of

equity. LnPRICE, LnEPS, LnBV, denote the logarithm of the respective variables. POLCON is Political Connection.

BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of

direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold.

Variable

LnEPS

LnBV

POLCON

BODFRANCO

BODINDO

LNSIZE

MANOWN

INSTOWN

LNAGE

CG2004

Adj R²

F-statistic

POLCON*EPS

POLCON*BV

Industries Fixed

5% and 10% level, respectively (one-tailed).

LEV

INTERCEPT

1 2

2

-0.586

-1.294

0.409

11.316

0.245

8.897

0.130

1.976

-0.309

-4.266

0.108

0.513

0.018

1.336

0.035

1.613

0.118

4.073

0.293

2.628

0.413

7.957

-0.122

-1.625

Yes

0.757

83.100

3

-0.556

-1.308

0.394

9.640

0.351

5.862

0.435

2.719

-0.320

-4.768

0.029

0.142

0.003

0.242

0.029

1.478

0.117

4.143

0.356

3.225

0.463

8.919

-0.122

-1.773 0.022

0.403

-0.180 -2.688

Yes

0.761

76.3740

Table 7: Main Regressions (2001-2016, n=541)

1

1.641

18.492

0.475

13.897

0.228

8.620

Expected Direction

?

+

?

+

+

+

+

+

+

?

Yes

0.727

This table presents the result after adjusted heteroscedasticity and t-statistics(in paratheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. *LnPRICE*, *LnEPS*, *LnBV*, denote the logarithm of the respective variables. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold. .***,**,* denote significance at 1%,

52

142.393

3
4
5
5
6
7
8
9
10
11
11
12
13
14
15
16
10
17
18
19
20
21
22
22
23
24
25
26
27
27
20
29
30
31
32
33
24
34
35
36
37
38
20
27
40
41
42
43
44
 1 E
4J
46
47
48
49
50
50
51
52
53
54
55
55
50

58
59
60

Commented [TM1]: To redo using log of price

T	able 8: Self-seled	ction Te	st			
Dependent variable	POLCON		PRICE		PRICE	
Variable	1		2		3	
INTERCEPT	0.305		-1.141		-2.055	
	1.072		-1.143		-1.712	*
BIG4	0.043					
	0.823					
LnEPS	0.028		0.434		0.459	
	2.022	**	7.385	***	7.150	***
LnBV	-0.024		0.227		0.310	
	-1.658	*	7.479	***	7.021	***
POLCON			0.131		0.477	
			1.983	**	2.634	**
BODFRANCO	0.260		-0.115		0.202	
	2.732	***	-0.370		0.519	
BODINDO	0.957		0.917		2.201	
	8.802	***	0.627		1.263	
LNSIZE	0.002		0.021		0.009	
	0.228		1.568		0.687	
LEV	-0.083		-0.037		-0.163	
	-5.770	***	-0.279		-1.033	
MANOWN	-0.062		0.065		-0.027	
	-6.396	***	0.671		-0.233	
INSTOWN	0.386		0.630		1.272	
	4.086	***	0.972		1.604	
LNAGE	-0.082		0.337		0.263	
	-1.617		2.139	**	1.613	
CG2004	-0.035		-0.148		-0.196	
	-0.470		-1.979	**	-2.567	**
IMILLS			0.858		2.316	
			0.548		1.234	
POLCON*LnEPS					0.018	
					0.328	
POLCON*LnBV					-0.197	, , , , , , , , , , , , , , , , , , ,
					-2.582	**
Industry fixed	Yes		Yes		Yes	
McFadden R ² / Adjusted R ²	0.338		0.757		0.761	
LR/F statistic	14.424	***	78.606	***	72.913	***
Obs with Dep=0	343					
Obs with Dep=1	198					

This table presents the result after adjusted heteroscedasticity and t-statistics (in paratheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity1. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. SIZE is total assets in millions of Mauritian Rupees. LNSIZE is the Natural log transformation of SIZE. LEV is total debt divided by total assets. MANOWN is the percentage of direct managerial ownership. INSTOWN is

<text> percentage ownership by the top five direct institutional shareholders. AGE is Natural log transformation of the sum of years since incorporation. Significant p-values are bold. ***, **, * denote significance at 1%, 5% and 10% level, respectively (one-tailed).

8 Table 9: Regressi	ons results on	Political connec	ctions and E	Ethnicit	y (2001-2	2016, n	=541)	
1 0 ⁄ariable	Expected							
11	Direction	1	2		3		4	
1 J ntercept	?	-0.586	-0.663		-0.703		-0.184	
13		-1.294	-1.604		-1.686	*	-0.441	
1 AnEPS	+	0.409	0.752		0.605		0.310	
15		11.316 ***	5.861	***	6.002	***	4.879	***
1 E ^{nBV}	+	0.245	-0.020		0.134		0.253	
17		8.897 ***	-0.146		1.214		3.282	***
POLCON	?	0.130	0.133		0.266		0.128	
10		1.976 *	1.971	**	1.251		0.617	
BODFRANCO	+	-0.309	-0.209		-0.319		-0.311	
20		-4.266 ***	-0.932	**	-1.406		-2.414	**
2 BODINDO	-	0.108	-0.206		-0.011		-1.597	
22		0.513	-0.597		-2.214	**	-2.309	***
230DFRANCO*LnEPS	+		-0.358		-0.310			
24			-2.092	**	-2.114	**		
25JODFRANCO*LnBV	+		0.189		0.252			
26			1.129		1.661	*	0.5(1	
27 ^{*ODINDO*LnEPS}	-		-0.586	-11-			0.761	*** ***
28			-3.741	***			2.082	**
29 ^{001NDO*LNBV}	-		0.511	** **			0.265	
30	0		2.449	**	0.255		0.055	
31	2				-0.333	***	0.512	***
32 POLCON*L nPV	9				-2.973		0.260	
33	1				2.065	**	2 8 20	***
34 OLCON *RODER ANCO	2				0.299		-2.020	
35					0.844			
36 OLCON *BODINDO	?				0.077		1 564	
37	•						2.123	**
38OLCON								
39 ^{*BODFRANCO*LnEPS}	?				0.639			
40					3.434	***		
POLCON *BODFRANCO*LnBV 41	?				-0.746			
42					-4.091	***		
POLCON *BODINDO*LnEPS	?						-1.251	
10							-3.320	***
TPOLCON *BODINDO*LnBV	?						0.248	
							0.585	
4 ⁴ Control Variables		Yes	Yes		Yes		Yes	
4 Industry Fixed		Yes	Yes		Yes		Yes	
$4 \otimes dj R^2$		0.757	0.768	<i></i>	0.778		0.785	
49 -statistic		83.100 ***	72.412	***	67.280	***	70.363	***

This table presents the result after adjusted heteroscedasticity and t-statistics(in parantheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors. Significant p-values are bold. ***, **, * denote significance at 1%, 5% and 10% level, respectively (one-tailed).

Table 10 Regression results for interaction between board ethnicity

	Bodfranco<	=med	Bodfranco>1	ned	Bodindo<=n	ned	Bodindo>m	ned
riable	1	- meu	2 2	neu	Doumdo < m	icu	Joundo- II	icu
liable	1		2		3		4	
TERCEPT	-0.188		3.280		-1.847		0.023	
	-0.301		3.272	***	-1.480		0.037	
EPS	0.407		0.203		1.421		0.298	
	3.825	***	3.154	***	4.244	***	2.959	***
BV	0.127		0.257		-0.426		0.323	
	0.901		2.989	***	-1.737	*	3.002	***
LCON	-0.407		0.622		0.248		0.367	
	-1.452		2.271	**	0.235		1.244	
DFRANCO	-0.778		-3.002		0.014		-1.416	
	-3.074	***	-4.828	***	0.053		-3.807	***
DINDO	-0.583		-8.872		0.492		0.370	
	-0.610		-3.975	***	0.380		1.338	
DINDO *LnEPS	-0.138		2.197					
	-0.318		2.252	**				
DINDO *LnBV	0.704		1.759					
	1.227		1.276					
DFRANCO *LnEPS					-1.244		0.058	
					-3.183	***	0.338	
DFRANCO *LnBV					0.808		0.553	
					3.111	***	2.559	**
DLCON*LnEPS	0.284		0.188		-0.480		-0.014	
	2.131	**	1.810	*	-1.043		-0.108	
DLCON*LnBV	0.009		-0.291		0.016		-0.101	
	0.051		-2.606	***	0.027		-0.621	
LCON*BODINDO	1.443		-2.564					
	1.383		-0.418		0.451		0.200	
LCON*BODFKANCO					0.451		-0.360	
	0 425		1 420		0.300		-0.341	
LCON BODINDO "LNEPS	-0.425		-1.438					
Ι ΓΟλΙ*ΡΟΠΙΝΠΟ *Ι ΡΙΖ	-0.939		-1.210					
LCON DODINDU "LNBV	-0.430		0./11					
Ι ΓΟΝΙ*ΡΟΠΕΡ ΑΝΓΟ *Ι «ΕΡΟ	-0.723		0.211		0 606		0.221	
LCON BODFRANCO LIEPS					0.090		0.251	
Ι ΓΟΝΙ*ΡΟΠΕΡ ΑΝΓΟ *Ι «ΡΙ					0.402		0.900	
LEON DODERANCO LADV					-0.402		-0.119	
ntrol variables	Vec		Vec		-0.045 Vec		-0.229 Vec	
hiror variables histry fired	Ves		Ves		Ves		Ves	
usu y Jurcu	279		262		282		259	
i R ²	0.865		0 732		0 767		0.836	
	0.005		0.152		0.707		0.050	

Provide the state of the sta This table presents the result after adjusted heteroscedasticity and t-statistics(in parantheses). PRICE is share price of firm i for the period t in Mauritian Rupees. EPS is earnings per share. BV is Book value per share of equity. POLCON is Political Connection. BODFRANCO is the percentage of Franco-Mauritian directors on the board of directors. BODINDO is the percentage of Indo-Mauritian directors from Indian descent on the board of directors.

Significant p-values are bold. ***, **, * denote significance at 1%, 5% and 10% level, respectively (one-tailed).